ON A HIGHER STRUCTURE ON OPERADIC DEFORMATION COMPLEXES

BORIS SHOIKHET

ABSTRACT. In this paper, we prove that there is a canonical homotopy (n+1)-algebra structure on the shifted operadic deformation complex $\text{Def}(e_n \to \mathcal{P})[-n]$ for any operad \mathcal{P} and a map of operads $f: e_n \to \mathcal{P}$. This result generalizes a result of Tamarkin, who considered the case $\mathcal{P} = \text{End}_{\mathcal{O}p}(X)$. Another more computational proof of the same result was recently sketched by Calaque and Willwacher.

Our method combines the one of Tamarkin, with the categorical algebra on the category of symmetric sequences, introduced by Rezk and further developed by Kapranov-Manin and Fresse. We define suitable deformation functors on *n*-coalgebras, which are considered as the "non-commutative" base of deformation, prove their representability, and translate properties of the functors to the corresponding properties of the representing objects. A new point, which makes the method more powerful, is to consider the argument of our deformation theory as an object of the category of symmetric sequences of dg vector spaces, not as just a single dg vector space .

1. Introduction

1.1. In the beautiful paper [T2], Dima Tamarkin proved that, for an algebra X over the operad \mathbf{e}_n , $n \geq 2$, the deformation complex $\operatorname{Def}(X)[-n]$ admits a natural structure of homotopy (n + 1)-algebra. Here $\mathbf{e}_n = H \cdot (E_n, \mathbb{k})$ is the homology operad of the *n*dimensional little discs operad E_n , and char $\mathbb{k} = 0$. In this paper, we consider \mathbf{e}_n as a *non-unitary* operad, in the sense of [F2], that is, $\mathbf{e}_n(0) = 0$.

If one considered the case n = 1 and took the operad Assoc for e_1 , the claim would be the famous Deligne conjecture for Hochschild cochains; all known proofs of it use transcendental methods such as Drinfeld associators.¹⁾ The result of [T2] is proven purely algebraically, without any transcendental methods. Moreover if one took the Poisson operad for e_1 , the method of [T2] would work as well.

Notice that the deformation complex $\operatorname{Def}_{\mathcal{O}}(X)$ of an algebra X over a Koszul operad \mathcal{O} is the same that the deformation complex of the corresponding map of operads $\operatorname{Def}(\mathcal{O} \to \operatorname{End}_{\mathcal{O}p}(X))$, where $\operatorname{End}_{\mathcal{O}p}(X)$ is the endomorphism operad of X, $\operatorname{End}_{\mathcal{O}p}(X)(n) =$ $\operatorname{Hom}_{\Bbbk}(X^{\otimes n}, X)$.

In this paper, we provide a proof of similar statement for the case of the general

Received by the editors 2017-08-31 and, in final form, 2018-10-08.

Transmitted by Clemens Berger. Published on 2018-10-14.

²⁰¹⁰ Mathematics Subject Classification: Primary: 13D10, Secondary: 18D50.

Key words and phrases: Deformation theory, operads, higher structures.

[©] Boris Shoikhet, 2018. Permission to copy for private use granted.

¹⁾See Remark 1.3 below.

deformation complex $\operatorname{Def}(\mathbf{e}_n \xrightarrow{f} \mathcal{P})$, where \mathcal{P} is an operad and f is a map of operads.

Throughout the paper, k denotes a field of characteristic 0. Denote by $\mathcal{V}ect(k)$ the k-linear abelian category of (unbounded) complexes of k-vector spaces.

We prove here the following statement:

1.2. THEOREM. For any operad \mathcal{P} in $Vect(\mathbb{k})$, and a morphism of operads $t: \mathbf{e}_n \to \mathcal{P}$, the shifted deformation complex

$$\operatorname{Def}(\mathbf{e}_n \xrightarrow{t} \mathcal{P})[-n]$$
 (1.1)

admits a natural structure of a homotopy (n + 1)-algebra. Its underlying homotopy $\text{Lie}\{-n\}$ -structure is strict and is given by the operadic convolution Lie algebra.²

For the case of deformation complex $\operatorname{Def}(\mathbf{e}_n \xrightarrow{\operatorname{id}} \mathbf{e}_n)[-n]$, the corresponding Lie bracket of degree -n is homotopically trivial, and the complex $\operatorname{Def}(\mathbf{e}_n \xrightarrow{\operatorname{id}} \mathbf{e}_n)[-n]$ becomes a homotopy (n+2)-algebra. We are going to discuss it elsewhere.

1.3. REMARK. Strictly speaking, the Deligne conjecture is the statement that there is an action of the chain operad $C(E_2, \Bbbk)$ on the Hochschild cochains. For this statement, several proofs which work over \mathbb{Q} without any transcendental methods are known, see [MS1,2], [BB], [BF], and [B1,2]+[T4]. On the other hand, a homotopy 2-algebra is an algebra over the operad hoe₂ which is the Koszul resolution of the operad \mathbf{e}_2 . Any known construction of quasi-isomorphism of operads hoe₂ $\rightarrow C(E_2, \Bbbk)$ uses transcendental methods.

1.4. When the paper had almost been completed, the author found it out that the recent paper [CW, Section 3] contains a sketch of another proof of Theorem 1.2, based on different ideas. The proof relies on a nice construction loc. cit., Section 3.1, providing an operadic twisting interpretation of the Kontsevich-Soibelman operad [KS, Sect. 5], what makes it possible to define its counterpart for any Hopf cooperad. The authors re-interpret the construction in [T2] as a map of graded operads $hoe_{n+1} \rightarrow Br_{n+1}$ (in the notations of [CW]). After that, everything reduces to a lengthy computation of the compatibility of this map with the differentials, loc.cit. Section 3.2 and Appendix A. Unfortunately, this computation was only sketched.

An advantage of our approach, compared with loc.cit., is being more conceptual and categorical, and not relying on computations. The author thinks that this paper, even though being just an account on another proof, has its own right for existence.

1.5. Let us outline the methods we employ to prove Theorem 1.2. We combine the methods of [T2] (see Section 1.6 below) with the "categorical algebra" on the category of symmetric sequences, developed in [R], [KM], [St].

The operads are defined as monoids in the category of symmetric sequences, with respect to the composition product. The composition product admits an inner Hom which is right adjoint with respect to the left factor, denoted by [-, -]. In particular,

²⁾For the case $\mathcal{P} = \operatorname{End}_{\mathcal{O}p}(X)$, this bracket can be thought of as the Gerstenhaber-like bracket.

for any symmetric sequence X, the inner Hom [X, X] is an operad; the operad $\operatorname{End}_{\mathcal{O}p}(V)$ for a vector space V is recovered as $[V^{(0)}, V^{(0)}]$ where the symmetric sequence $V^{(0)}$ is $V^{(0)}(0) = V$ and $V^{(0)}(n) = 0$ for $n \neq 0$.

One can talk on left modules over an operad \mathcal{O} in the category of symmetric sequences; it is a symmetric sequence X with a map $\mathcal{O} \circ X \to X$ which is associative and such that $id \in \mathcal{O}(1)$ acts as identity on X. A conventional algebra V over \mathcal{O} is recovered as the case of the symmetric sequence $V^{(0)}$ defined just above. To give a left \mathcal{O} -module structure on a symmetric sequence X is the same as to give an operad maps $\mathcal{O} \to [X, X]$.

One can as well talk on right Q-modules, where Q is an operad. In contrast with the left Q-modules, the right Q-modules form a k-linear abelian category. For two right Q-modules X, Y, one can define the relative internal hom $[X, Y]_Q$, and develop the corresponding "categorical algebra". In particular, $[X, X]_Q$ becomes an operad. It is due to [R], with subsequent development made in [KM], [F1], [St].

If a symmetric sequence X is an $\mathcal{O}-\mathcal{Q}$ -bimodule, one gets a map of operads

$$\mathcal{O} \to [X, X]_{\mathcal{Q}} \tag{1.2}$$

Our first goal is to extend the result of Tamarkin to a map of operad $f : \mathbf{e}_n \to [X, X]$, where X is a symmetric sequence (originally it was proven for the case when X has only arity 0 non-zero component). It does not meet any trouble. We mention that the barcomplex $\operatorname{Bar}_{\mathcal{O}}(X)$ is defined as a *symmetric sequence* with a component-wise differential.

As the next step, we take a $\mathbf{e}_n - \mathcal{Q}$ -bimodule X, and extend the Tamarkin theory [T2] for the corresponding operad morphism $\mathbf{e}_n \to [X, X]_{\mathcal{Q}}$, see (1.2). To this end, we prove the " \mathcal{Q} -relative" version of the classical statement, expressing maps of dg coalgebras $C \to \text{Bar}_{\mathcal{O}}(X)$ as the Maurer-Cartan elements in the convolution Lie algebra $\text{Hom}(C/\Bbbk, X)[-1]$ (here \mathcal{O} is a Koszul operad such that the cooperad \mathcal{O}^i is biaugmented, and the \mathcal{O}^i coalgebra C is pro-conilpotent, as well as for the classical case³), see Section 4.7.4.

Then we apply it to the case $X = \mathcal{P}$, considered as a right module over $\mathcal{Q} = \mathcal{P}$. A map of operads $f: \mathcal{O} \to \mathcal{P}$ makes \mathcal{P} an $\mathcal{O}-\mathcal{P}$ -bimodule. One has:

$$[\mathcal{P}, \mathcal{P}]_{\mathcal{P}} = \mathcal{P} \tag{1.3}$$

as an operad, and the map (1.2)

$$\mathcal{O} \to [\mathcal{P}, \mathcal{P}]_{\mathcal{P}} = \mathcal{P} \tag{1.4}$$

is equal to f.

Thus, any map of operads appears as a case of the map (1.2), and the general statement follows from the case of operad maps (1.2).

 $^{^{3)}\}mathrm{See}$ Section 3.1 for the definitions of a bi-augmented cooperad and a pro-conilpotent coalgebra over it

1.6. Let us briefly outline the main ideas of [T2], referring the reader to the Introduction to loc.cit. for a more detailed overview.

First of all, the deformation complex $\operatorname{Def}_{\mathcal{O}}(X) = \operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X)[1]$, where the r.h.s. is the deformation complex of the identity morphism of an \mathcal{O} -algebra X. One can consider more general deformation complexes $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{f} Y)$ of a morphism of \mathcal{O} -algebras, what provides a relative version of the initial deformation complex. The convention we adapt here is that all deformation complexes we deal with are considered with the grading making them a dg Lie algebra, that is, the underlying Lie bracket preserves the grading.

There is the following conceptual way to think on the deformation complex $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{f} Y)$. Y). One associates with a morphism $f: X \to Y$ of algebras over a Koszul operad \mathcal{O} a functor $F_{X,Y}^f$ on coaugmented cocommutative coalgebras with values in sets:

$$F_{X,Y}^f(a) = \{ \phi \in \operatorname{Hom}_{\operatorname{Coalg}(\mathcal{O}^{\mathsf{i}})}(a \otimes \operatorname{Bar}_{\mathcal{O}}(X), \operatorname{Bar}_{\mathcal{O}}(Y)), \phi \circ (\eta \otimes \operatorname{id}_{\operatorname{Bar}_{\mathcal{O}}}) = \operatorname{Bar}(f) \}$$
(1.5)

where $\eta: \mathbb{k} \to a$ is the coaugmentation map. The bar-complexes are coalgebras over the (shifted) Koszul dual cooperad \mathcal{O}^i , and Hom is taken in the category of \mathcal{O}^i -coalgebras. The map f defines a map $\operatorname{Bar}(f): \operatorname{Bar}_{\mathcal{O}}(X) \to \operatorname{Bar}_{\mathcal{O}}(Y)$ of \mathcal{O}^i -coalgebras, and one considers the maps $\phi: a \otimes \operatorname{Bar}_{\mathcal{O}}(X) \to \operatorname{Bar}_{\mathcal{O}}(Y)$ of \mathcal{O}^i -coalgebras, equal to $\operatorname{Bar}(f)$ on the "special (co)fiber". Notice that, if one worked with the Schlessinger framework of deformation theory, one would consider the functor (on artinian coalgebras), defined as $a \mapsto \{t \in \operatorname{Hom}_{\mathcal{Alg}(\mathcal{O})}(\operatorname{Hom}_{\mathbb{k}}(a, X), Y), t \circ \eta = f\}$. The functor $F_{X,Y}^f$ should be considered as a derived version of the latter functor; working with the Deligne-Drinfeld framework of deformation theory (which describes the deformations via the Maurer-Cartan elements in a dg Lie algebra), one should replace the functors themselves with their "derived versions".

If one restricts ourselves with a full subcategory of the category of cocommutative coalgebras called pro-conilpotent (aka connected in [Q], aka pro-artinian in [T2]), and assumes that the cooperad \mathcal{O}^{i} is biaugmented, the functor $F_{X,Y}^{f}$ becomes representable:

$$F_{X,Y}^f(a) = \operatorname{Hom}_{\operatorname{Coalg}}(a, a_{\operatorname{rep}}(f))$$

for some pro-conilpotent cocommutative coalgebra $a_{rep}(f)$, where Coalg stands for the category of pro-conilpotent cocommutative coalgebras over k. Tamarkin shows that

$$a_{\rm rep}(f) = C_{\rm CE}({\rm Def}_{\mathcal{O}}(X \xrightarrow{f} Y), \Bbbk)$$

where $C_{\text{CE}}(-)$ stands for the Chevalley-Eilenberg chain complex of a dg Lie algebra. It defines the dg Lie algebra $\text{Def}_{\mathcal{O}}(X \xrightarrow{f} Y)$ up to a quasi-isomorphism. The Lie bracket is the algebra-convolution Lie bracket, see Section 2.13.1.

Then the composition property of the functor $F_{X,Y}^f$, saying that, for two maps

$$X \xrightarrow{f} Y \xrightarrow{g} Z$$

one has a (functorial) map

$$F_{Y,Z}^g(a') \times F_{X,Y}^f(a) \to F_{X,Z}^{gf}(a' \otimes a)$$
(1.6)

is translated to the corresponding property of the representing coalgebras:

$$a_{\rm rep}(f) \otimes a_{\rm rep}(g) \to a_{\rm rep}(gf)$$
 (1.7)

It is associative, for a chain of 3 morphisms. It follows that, for $f = id : X \to X$, one has a monoid structure

$$a_{\rm rep}(X) \otimes a_{\rm rep}(X) \to a_{\rm rep}(X)$$

where $a_{rep}(X) = a_{rep}(id_X)$.

The conclusion is that $a_{\text{rep}}(X) = C_{\text{CE}}(\text{Def}_{\mathcal{O}}(X \xrightarrow{\text{id}} X))$ becomes a cocommutative dg bialgebra (whose underlying cocommutative coalgebra is cofree pro-conilpotent). Then the Milnor-Moore theorem applied to the underlying bialgebra $a_{\text{rep}}(X)$ (with forgotten differential) implies that the convolution Lie bracket on $\text{Def}_{\mathcal{O}}(X \xrightarrow{\text{id}} X)$ vanishes, and the shifted complex $\text{Def}_{\mathcal{O}}(X \xrightarrow{\text{id}} X)[1]$ gets a Lie bracket.

Indeed, the Milnor-Moore theorem says that $a_{\rm rep}(X)$ is the pro-conilpotent universal enveloping (co)algebra $\mathcal{U}(\mathfrak{g})$, where \mathfrak{g} is the graded space of primitive elements in the coalgebra $a_{\rm rep}(X)$. One easily identifies \mathfrak{g} with $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X)[1]$. The quadratic component of the Chevalley-Eilenberg differential on \mathfrak{g} vanishes by definition. Now the Leibniz rule for the dg bialgebra $C_{\rm CE}(\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X))$ shows that the quadratic component of the Chevalley-Eilenberg differential vanishes on the entire complex, therefore, the convolution Lie bracket on $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X)$ vanishes. On the other hand, $\mathfrak{g} = \operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X)[1]$ gets a new Lie bracket, $[a, b] = a * b - (-1)^{|a||b|}b * a$, where a, b are primitive, and -*- is the product on $a_{\rm rep}(X)$ (given by the monoid structure on $a_{\rm rep}(\operatorname{id}_X)$, as above). This Lie bracket is clearly compatible with the differential on $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{\operatorname{id}} X)[1]$, because the product -*- is.

It is the Lie bracket of Gerstenhaber $type^{4}$.

Note that for $f \neq id$, the algebra convolution Lie bracket on $\operatorname{Def}_{\mathcal{O}}(X \xrightarrow{f} Y)$ is not 0. Therefore, the above construction can be thought of as a sort of "quasi-classical limit".

Notice that what makes all these constructions possible, is the fact that, for any cooperad \mathcal{C} and a \mathcal{C} -coalgebra B, the tensor product $a \otimes B$ with a cocommutative coalgebra a is again a \mathcal{C} -coalgebra. On the language of operads, it follows from existence of a map of cooperads $\mathsf{Comm}^* \otimes \mathcal{C} \to \mathcal{C}$ (which holds as $\mathsf{Comm}^* = \mathsf{Comm}$ is the unit for the product \otimes in symmetric sequences, so $\mathsf{Comm}^* \otimes \mathcal{C} = \mathcal{C}$).

A new fundamental idea in [T2] was to take for a higher structured algebraic objects than cocommutative coalgebras. That is, D.Tamarkin considers "deformations with noncommutative base". Namely, he considers the case when $\mathcal{C} = \mathcal{O}^{i} = \mathbf{e}_{n}^{*}$, the dual cooperad to the operad \mathbf{e}_{n} . The operad \mathbf{e}_{n} is a Hopf operad, that is, there is an operad map

$$\mathbf{e}_n \to \mathbf{e}_n \otimes \mathbf{e}_n \tag{1.8}$$

⁴)Here we mean that the Lie bracket is defined similarly with the Gerstenhaber bracket on Hochschild cochain complex

and thus, for the dual cooperad one has

$$\mathbf{e}_n^* \otimes \mathbf{e}_n^* \to \mathbf{e}_n^* \tag{1.9}$$

(we call it Hopf cooperad). (In fact, the coproduct like (1.8) exists for any operad in graded vector spaces defined as the homology operad of a topological operad. Indeed, for the topological spaces there is the arity-wise diagonal map. Passing to homology, it indices a coproduct).

It follows from (1.8) that the tensor product of two \mathbf{e}_n -algebras is again a \mathbf{e}_n -algebra, and it follows from (1.9) that the tensor product of two \mathbf{e}_n^* -coalgebras is again a \mathbf{e}_n^* -coalgebra.

The case $C = O^{i} = \mathbf{e}_{n}^{*}$ is corresponded to $O = \mathbf{e}_{n}\{n\}$ (by $O\{1\}$ is denoted the operadic shift, see Section 2.8.1).

It follows that for $\mathcal{O} = \mathbf{e}_n\{n\}$ one can upgrade the functor $F_{X,Y}^f$ to another functor $G_{X,Y}^f$ defined exactly as (1.5), but with $a \neq a^*$ -coalgebra. So $G_{X,Y}^f$ is a functor from the category $\mathcal{C}oalg_n$ of pro-conilpotent *n*-coalgebras to the category $\mathcal{S}ets$.

The functor $G_{X,Y}^{f}$ is also representable, and the representing object is

$$A_{\operatorname{rep}}(f) = \operatorname{Bar}_{\mathbf{e}_n\{n\}}(\operatorname{Def}(X \xrightarrow{f} Y)[1])$$

where $\operatorname{Def}(X \xrightarrow{f} Y)[1]$ is endowed with a $e_n\{n\}$ -algebra structure. It has the property similar to (1.6), translated to a map (1.7) of *n*-coalgebras. It implies that $A_{\operatorname{rep}}(X) := A_{\operatorname{rep}}(\operatorname{id}_X)$ is a monoid object in the category of *n*-coalgebras. Note that X may be a $e_n\{k\}$ -algebra for any k, as, by an operadic shift, a map of $e_n\{k\}$ -algebras can be made a map of $e_n\{n\}$ algebras, as the construction requires.

Moreover, there is an imbedding $i: \operatorname{Coalg} \to \operatorname{Coalg}_n$, and the restriction of the functor $G_{X,Y}^f$ along it is equal to $F_{X,Y}^f$. The functor i admits right adjoint, what makes possible to link the representing objects $a_{\operatorname{rep}}(X)$ and $A_{\operatorname{rep}}(X)$, along with their monoid structures.

These constructions provide us with some higher structure on the deformation complex $\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)$, where X is an \mathbf{e}_n -algebra. Having unwound this structure, D.Tamarkin derived a homotopy (n+1)-algebra structure on $\operatorname{Def}_{\mathbf{e}_n}(X)[-n]$.

To be more precise, to get a homotopy \mathbf{e}_{n+1} algebra structure on $\operatorname{Def}_{\mathbf{e}_n}(X)[-n]$, one uses solely the deformation functor $G_{X,Y}^f$, corresponded to the deformation theory with \mathbf{e}_n coalgebra base, and the monoid structure on the $\mathbf{e}_n\{n\}$ coalgebra $A_{\operatorname{rep}}(X)$, see Section 6.1. The comparison of the two deformation theories, given by the functors $F_{X,Y}^f$ and $G_{X,Y}^f$, via the right adjoint R to the functor $i: \operatorname{Coalg} \to \operatorname{Coalg}_n$, is employed to get an explicit description of the underlying homotopy Lie bracket on $\operatorname{Def}_{\mathbf{e}_n}(X)$ of the \mathbf{e}_{n+1} algebra structure on $\operatorname{Def}_{\mathbf{e}_n}(X)[-n]$. Namely, this bracket is identified with the operadic convolution (strict) Lie bracket. The functor i is (colax-)monoidal, therefore, its right adjoint is lax monoidal. As such, it sends monoids to monoids. The key point is to show that the right adjoint functor R sends $A_{\operatorname{rep}}(X)$ to $a_{\operatorname{rep}}(X)$ with their monoid structures, see Section 6.2.

1.7. ORGANISATION OF THE PAPER. The paper is organized as follows.

In Section 2 we recall some facts on operads which will be used in the paper. Nothing here is new.

In Section 3 we recall the bar-cobar duality, for algebras X over a Koszul operad \mathcal{O} . We extend the classical statement to the case when X is a symmetric sequence, that is, is a left module over \mathcal{O} . The bar and cobar complexes become symmetric sequences with component-wise differentials, equivariant with respect to the action of the symmetric groups. In Section 3.5 we consider the situation when X is a $\mathcal{O}-\mathcal{Q}$ -bimodule, for an operad \mathcal{Q} . Then the bar-complex is at once a coalgebra over the cooperad \mathcal{O}^{i} and a right module over the operad \mathcal{Q} . We prove a fragment of the corresponding bar-cobar duality statement, for this setting.

Section 4 contains the deformation theory with cocommutative cobase. We start with the case when the target operad \mathcal{P} of the map $f : \mathbf{e}_n \to \mathcal{P}$, we consider the deformations of, is [X, X], for a symmetric sequence X. Then we consider the case $\mathcal{P} = [X, X]_{\mathcal{Q}}$, for X a $\mathbf{e}_n - \mathcal{Q}$ -bimodule. As we have shown above, this case covers an arbitrary \mathcal{P} .

In Section 5, we consider the deformation theory with e_n -coalgebra as the cobase. We extend the results of Section 4 to this context.

Finally, Section 6 contains a derivation of Theorem 1.2 from the previous results. This part is very similar to [T2, Section 5], though we tried to clarify some points.

ACKNOWLEDGEMENTS. The author is thankful to Michael Batanin, Anton Gerasimov, and Dima Tamarkin, for valuable discussions. The author is thankful to the two anonymous referees for their careful reading of the paper and many remarks and corrections, which have undoubtedly made this published version more readable.

The work was partially supported by the FWO Research Project Nr. G060118N and by the Russian Academic Excellence Project '5-100'.

2. Preliminaries

Throughout the paper, k is a field of characteristic 0. We denote by $\operatorname{Vect}(k)$ the symmetric monoidal category of (\mathbb{Z} -graded) dg vector spaces over k. We use the cohomological grading, so all differentials are of degree +1. For a (dg) vector space V with an action of a group G, we denote by V^G the subspace of invariants, and by V_G the quotient-space of coinvariants.

2.1. Operads.

2.1.1. SYMMETRIC SEQUENCES. A (symmetric) operad \mathcal{P} in $\mathcal{V}ect(\Bbbk)$ is a collection of Σ_n -representations $\mathcal{P}(n), n \geq 0$, with the operadic composition operation

$$\Upsilon \colon \mathcal{P}(k) \otimes \mathcal{P}(n_1) \otimes \cdots \otimes \mathcal{P}(n_k) \to \mathcal{P}(n_1 + \cdots + n_k)$$
(2.1)

and an element $id \in \mathcal{P}(1)$, such that the operadic composition is compatible with the actions of symmetric groups and is associative (in some natural sense), and id is a two-sided unit for the operadic composition.

Alternatively, one can define an operad a monoid in the monoidal category of *symmetric sequences*, with respect to the composition product.

By definition, a symmetric sequence in a category \mathcal{C} is a collection of objects $\{\mathcal{P}(n) \in \mathcal{C}\}_{n\geq 0}$ with an action of symmetric group Σ_n on $\mathcal{P}(n)$. A morphism $\phi: \mathcal{P} \to \mathcal{Q}$ of symmetric sequences is a collection of Σ_n -equivariant maps $\{\mathcal{P}(n) \to \mathcal{Q}(n)\}_{n\geq 0}$:

$$\operatorname{Hom}_{\Sigma}(\mathcal{P}, \mathcal{Q}) = \prod_{n \ge 0} \operatorname{Hom}_{\Bbbk}(\mathcal{P}(n), \mathcal{Q}(n))^{\Sigma_n}$$
(2.2)

We denote the category of symmetric sequences in \mathcal{C} with the morphisms $\operatorname{Hom}_{\Sigma}(-,-)$ by \mathcal{C}_{Σ} .

2.2. DEFINITION. A dg symmetric sequence is a symmetric sequence \mathcal{P} in $\mathcal{V}ect(\Bbbk)_{\Sigma}$ (that is, each $\mathcal{P}(n)$ is a complex of \Bbbk -vector spaces), such that the differential on $\mathcal{P}(n)$ commutes with the action of Σ_n on it, for each $n = 0, 1, 2, \ldots$

It is a monoidal category, with the monoidal product $\mathcal{P} \circ \mathcal{Q}$ defined as

$$(\mathcal{P} \circ \mathcal{Q})(n) = \bigoplus_{k \ge 0} \mathcal{P}(k) \otimes_{\Sigma_k} \mathcal{Q}^{\boxtimes k}(n)$$
(2.3)

where

$$\mathcal{Q}^{\boxtimes k}(n) = \bigoplus_{n_1 + \dots + n_k = n} \operatorname{Ind}_{\Sigma_{n_1} \times \dots \times \Sigma_{n_k}}^{\Sigma_n} (\mathcal{Q}(n_1) \otimes \dots \otimes \mathcal{Q}(n_k))$$
(2.4)

This product on the category of symmetric sequences is called *the composition product*, it is associative:

$$(\mathcal{P} \circ \mathcal{Q}) \circ \mathcal{R} = \mathcal{P} \circ (\mathcal{Q} \circ \mathcal{R}) \tag{2.5}$$

Its unit is the symmetric sequence I, defined as I(1) = k, I(n) = 0 for $n \neq 1$.

2.3. REMARK. Note that for a group G and two G-modules V and W, the tensor product $V \otimes_G W$ means the space of *coinvariants* $(V \otimes_{\Bbbk} W)_G$. That is, in (2.3) one takes the *coinvariants* with respect to the group Σ_k action.

One defines version(s) of (2.3) for *invariants* instead of coinvariants:

$$(\mathcal{P} \circ_1 \mathcal{Q})(n) = \bigoplus_{k \ge 0} \left(\mathcal{P}(k) \otimes \mathcal{Q}^{\boxtimes k}(n) \right)^{\Sigma_k}$$
(2.6)

$$(\mathcal{P}\hat{\circ}_1 \mathcal{Q})(n) = \prod_{k \ge 0} \left(\mathcal{P}(k) \otimes \mathcal{Q}^{\boxtimes k}(n) \right)^{\Sigma_k}$$
(2.7)

These products are monoidal products in $\mathcal{V}ect(\mathbb{k})_{\Sigma}$, with the same unit I.

One re-interprets the concept of operad saying that an operad is a monoid in the category of symmetric sequences, with respect to the composition product. That is, a symmetric sequence \mathcal{P} is an operad, if there are maps

$$\Upsilon \colon \mathcal{P} \circ \mathcal{P} \to \mathcal{P}, \quad i \colon I \to \mathcal{P} \tag{2.8}$$

satisfying the axioms of a monoid in a monoidal category. An operad \mathcal{P} such that $\mathcal{P}(n) = 0$ unless n = 1, $\mathcal{P}(1) = A$ is the same as an associative algebra. More generally, consider the functor

$$i: \mathcal{V}ect(\mathbb{k}) \to \mathcal{V}ect(\mathbb{k})_{\Sigma}, \quad V \mapsto X_V = (0, \bigcup_{n=1}^V, 0, 0, \dots)$$

It is a strict monoidal functor

$$i \colon (\mathcal{V}ect(\Bbbk), \otimes) \to (\mathcal{V}ect(\Bbbk)_{\Sigma}, \circ)$$

A cooperad in $\operatorname{Vect}(\mathbb{k})$ is a comonoid in $\operatorname{Vect}(\mathbb{k})_{\Sigma}$ with respect to the monoidal product $-\hat{\circ}_1-$. That is, a cooperad is a symmetric sequence \mathcal{C} with maps

$$\Theta \colon \mathcal{C} \to \mathcal{C} \hat{\circ}_1 \mathcal{C}, \quad \varepsilon \colon \mathcal{C} \to I \tag{2.9}$$

satisfying the usual comonoid axioms.

A cooperad is called *finite* if the structure map (2.9) factors as

$$\mathcal{C} \to \mathcal{C} \circ_1 \mathcal{C} \to \mathcal{C} \hat{\circ}_1 \mathcal{C} \tag{2.10}$$

The composition product $-\circ$ - admits an inner hom, defined as

$$[\mathcal{P}, \mathcal{Q}](n) = \operatorname{Hom}_{\Sigma}(\mathcal{P}^{\boxtimes n}, \mathcal{Q})$$
(2.11)

It is right adjoint to $-\circ -$:

$$\operatorname{Hom}_{\Sigma}(\mathcal{P} \circ \mathcal{Q}, \mathcal{R}) = \operatorname{Hom}_{\Sigma}(\mathcal{P}, [\mathcal{Q}, \mathcal{R}])$$
(2.12)

For any symmetric sequence X, [X, X] is a monoid with respect to \circ , thus an operad.

An important particular case is when X(n) = 0 unless n = 0, X(0) = W. Then $[X, X](n) = \operatorname{Hom}_{\Bbbk}(W^{\otimes n}, W)$. It is an operad with the operadic composition defined by plugging the arguments. We denote this operad by $\operatorname{End}_{\mathfrak{O}p}(X)$.

For any group G and a G-module M, there is a map from invariants to coinvariants:

$$can\colon M^G \to M_G \tag{2.13}$$

In particular, for any two symmetric sequences X, Y, one has a canonical map

$$can_{X,Y} \colon X \circ_1 Y \to X \circ Y \tag{2.14}$$

If the group G is finite, and the order $\sharp G$ is invertible in \Bbbk , the map (2.13) is an isomorphism, with the inverse given by the norm:

$$N: M_G \to M^G$$

$$m \mapsto \frac{1}{\sharp G} \sum_{g \in G} g \circ m \tag{2.15}$$

In this paper, char $\mathbf{k} = 0$, and the map (2.14) is always an isomorphism.

We often identify the products $-\circ -$ and $-\circ_1 -$, assuming these isomorphisms.

2.3.1. Along with the composition monoidal product on the category $\mathcal{V}ect(\Bbbk)_{\Sigma}$, we will consider the *level-wise monoidal product* $-\otimes_{lev}$ -, defined as:

$$(X \otimes_{lev} Y)(n) = X(n) \otimes Y(n)$$
(2.16)

with the diagonal action of the symmetric group Σ_n . It is called the Hadamard product and is denoted by $-\otimes_H -$ in [LV].

The unit for this product is the symmetric sequence Comm, defined as

$$\mathsf{Comm}(n) = \Bbbk, n \ge 0 \tag{2.17}$$

This product admits the internal Hom defined as

$$Hom_{lev}(X,Y) = \{Hom_{k}(X(n),Y(n))\}_{n \ge 0}$$
(2.18)

with the symmetric group acting as $(\sigma * f)(x) = \sigma(f(\sigma^{-1}x))$. It is right adjoint to $-\otimes_{lev}$ -:

$$\operatorname{Hom}_{\Sigma}(X \otimes_{lev} Y, Z) = \operatorname{Hom}_{\Sigma}(X, \operatorname{Hom}_{lev}(Y, Z))$$
(2.19)

In fact, one has a stronger adjunction:

$$\operatorname{Hom}_{lev}(X \otimes_{lev} Y, Z) = \operatorname{Hom}_{lev}(X, \operatorname{Hom}_{lev}(Y, Z))$$
(2.20)

2.4. LEMMA. Let X, Y, X_1, Y_1 be symmetric sequences in $\operatorname{Vect}(\mathbb{k})$. One has the following 4-functorial maps:

(i)

 $\eta_{X_1X_2Y_1Y_2} \colon (X_1 \otimes_{lev} X_2) \circ (Y_1 \otimes_{lev} Y_2) \to (X_1 \circ Y_1) \otimes_{lev} (X_2 \circ Y_2)$ (2.21)

(ii)

$$\mu_{X_1 X_2 Y_1 Y_2} \colon (X_1 \circ_1 Y_1) \otimes_{lev} (X_2 \circ_1 Y_2) \to (X_1 \otimes_{lev} X_2) \circ_1 (Y_1 \otimes_{lev} Y_2)$$
(2.22)

PROOF. We use the fact that for a group G and two G-modules there are maps

$$V^G \otimes W^G \to (V \otimes W)^G$$
 and $(V \otimes W)_G \to V_G \otimes W_G$ (2.23)

making the invariants functor a lax-monoidal functor, and the coinvariants functor a colax-monoidal functor.

Denote

$$(\mathcal{P} \star \mathcal{Q})(n) = \bigoplus_{k \ge 0} \mathcal{P}(k) \otimes_{\Bbbk} \mathcal{Q}^{\boxtimes k}(n)$$
(2.24)

and consider the r.h.s. as a Σ_k -module.

There are maps (the imbedding i and the projection p):

$$i: (X_1 \otimes_{lev} X_2) \star (Y_1 \otimes_{lev} Y_2) \to (X_1 \star Y_1) \otimes_{lev} (X_2 \star Y_2)$$

$$(2.25)$$

$$p: (X_1 \star Y_1) \otimes_{lev} (X_2 \star Y_2) \to (X_1 \otimes_{lev} X_2) \star (Y_1 \otimes_{lev} Y_2)$$
(2.26)

Both are maps of Σ_k -modules. The statement follows from (2.23).

2.5. COROLLARY. The level-wise tensor product of two operads is an operad. The levelwise tensor product of two finite cooperads is a (finite) cooperad.

Another important fact is

2.6. PROPOSITION. For any four symmetric sequences X, Y, X_1, Y_1 in $\operatorname{Vect}(\mathbb{k})$, there are 4-functorial maps

$$\operatorname{Hom}_{lev}(X,Y) \circ \operatorname{Hom}_{lev}(X_1,Y_1) \to \operatorname{Hom}_{lev}(X \circ X_1, Y \circ Y)$$

$$(2.27)$$

and

$$\operatorname{Hom}_{lev}(X \circ X_1, Y \circ Y) \to \operatorname{Hom}_{lev}(X, Y) \circ \operatorname{Hom}_{lev}(X_1, Y_1)$$
(2.28)

2.7. COROLLARY. Let \mathcal{P} be an operad and let \mathcal{C} be a finite cooperad in $\operatorname{Vect}(\Bbbk)$, char $\Bbbk = 0$. Then

 $\operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P})$

is naturally an operad, and

 $\operatorname{Hom}_{lev}(\mathcal{P},\mathcal{C})$

is naturally a finite cooperad.

2.7.1. LEFT AND RIGHT MODULES, BIMODULES. A left (resp., right) module over an operad \mathcal{P} is a symmetric sequence M with a map $\mathcal{P} \circ M \to M$ (resp., $M \circ \mathcal{P} \to M$) satisfying the usual module axioms.

Let us stress an essential difference between the left and the right modules over an operad \mathcal{P} : the right modules over \mathcal{P} always form a k-linear abelian category (in the dg situation, we upgrade it to the corresponding dg category over k), whereas the category of left \mathcal{P} -modules is even non-additive.

An algebra X over an operad \mathcal{P} is a symmetric sequence $X \in \mathcal{V}ect(\mathbb{k})_{\Sigma}$ equipped with a map of operads $\mathcal{P} \to [X, X]$. Alternatively, it is the same that a left \mathcal{P} -module structure on X:

$$m\colon \mathcal{P} \circ X \to X \tag{2.29}$$

We see from the adjunction (2.12) that X is an algebra over an operad \mathcal{P} iff it is a left module over \mathcal{P} .

When X(n) = 0 unless $n \neq 0$, $X(0) = X_0 \in \mathcal{V}ect(\mathbb{k})$, we call X_0 a *conventional* algebra over the operad \mathcal{P} .

For an operad \mathcal{P} and $A \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ consider

$$\mathcal{P}\langle A \rangle = \mathcal{P} \circ A \tag{2.30}$$

It follows from the associativity (2.5) that $\mathcal{P}\langle A \rangle$ is an algebra over \mathcal{P} . The functor $A \rightsquigarrow \mathcal{P}\langle A \rangle$ is left adjoint to the forgetful functor from algebras over \mathcal{P} to $\mathcal{V}ect(\Bbbk)_{\Sigma}$.

Let X be a symmetric sequence with a left action of the operad \mathcal{P} . Then $X \circ \mathbb{k}^{(0)} = X\langle \mathbb{k} \rangle$ is an algebra over \mathcal{P} . One has explicitly:

$$X \circ \mathbb{k}^{(0)} = \bigoplus_{n \ge 0} X(n)_{\Sigma_n} \tag{2.31}$$

Let $\mathcal{P}_1, \mathcal{P}_2$ be operads. A $(\mathcal{P}_1, \mathcal{P}_2)$ -bimodule is a symmetric sequence \mathcal{M} , with a left module structure over \mathcal{P}_1 and a right module structure over \mathcal{P}_2 which commute.

2.7.2. RELATIVE COMPOSITION PRODUCT AND THE RELATIVE INNER HOM. Here we outline some categorical algebra in the category $\mathcal{V}ect(\Bbbk)_{\Sigma}$. We refer the reader to [R], [KM], [St] for more detail.

Let \mathcal{P} be an operad in $\mathcal{V}ect(\Bbbk)$. Then right \mathcal{P} -modules form a dg category over \Bbbk , denoted by $\operatorname{Mod}-\mathcal{P}$. We denote by $\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(M,N)$ the \Bbbk -vector space of morphisms of symmetric sequences respecting the right \mathcal{P} -module structure.

For $X \in \mathcal{V}ect(\Bbbk)_{\Sigma}$, $Y \in Mod - \mathcal{P}$, the composition product $X \circ Y$ is a right \mathcal{P} -module. Define the *relative inner hom* functor

$$[-,-]_{\mathcal{P}} \colon (\mathrm{Mod}-\mathcal{P})^{\mathrm{op}} \times \mathrm{Mod}-\mathcal{P} \to \mathcal{V}ect(\Bbbk)_{\Sigma}$$

as follows. The symmetric sequence $[M, N]_{\mathcal{P}}$ is defined as the equalizer

$$[M,N]_{\mathcal{P}} \to [M,N] \stackrel{u}{\underset{v}{\Rightarrow}} [M \circ \mathcal{P},N]$$
(2.32)

where, for $f \in [M, N]$, $u(f) = \mu_N(f \circ \mathcal{P})$, and $v(f) = f \mu_M^{\boxtimes n}$, so that

$$[M, N]_{\mathcal{P}}(n) = \operatorname{Hom}_{\Sigma, \operatorname{Mod}-\mathcal{P}}(M^{\boxtimes n}, N)$$

There is a natural isomorphism

$$\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(X \circ M, N) = \operatorname{Hom}_{\Sigma}(X, [M, N]_{\mathcal{P}})$$
(2.33)

In this way, the category Mod $-\mathcal{P}$ becomes a monoidal category tensored and enriched over the monoidal category ($\operatorname{Vect}(\mathbb{k})_{\Sigma}, \circ, I$), cf. [F1, Ch.1].

In particular, for $M \in \text{Mod}-\mathcal{P}$, $[M, M]_{\mathcal{P}}$ is an operad. Moreover, for $M, N \in \text{Mod}-\mathcal{P}$, $[M, N]_{\mathcal{P}}$ is an $[N, N]_{\mathcal{P}}-[M, M]_{\mathcal{P}}$ -bimodule, and

$$[\mathcal{P}, \mathcal{P}]_{\mathcal{P}} = \mathcal{P} \tag{2.34}$$

as an $\mathcal{P}-\mathcal{P}$ -bimodule, and

$$[\mathcal{P}, M]_{\mathcal{P}} = M \tag{2.35}$$

as right \mathcal{P} -module.

One can define the relative composition product $M \circ_{\mathcal{P}} N \in \mathcal{V}ect(\mathbb{k})_{\Sigma}$, for $M \in Mod-\mathcal{P}$, $N \in \mathcal{P}-Mod$, as the corresponding coequalizer.

Moreover, for M an A-B-bimodule, N a B-C-bimodule, one defines $M \circ_B N \in A-Mod-C$. It gives rise to a functor

$$-\circ_B -: A - \operatorname{Mod} - B \times B - \operatorname{Mod} - C \to A - \operatorname{Mod} - C$$
 (2.36)

One has the adjunction:

$$\operatorname{Hom}_{\Sigma,A-\operatorname{Mod}-C}(M \circ_B N, L) = \operatorname{Hom}_{\Sigma,A-\operatorname{Mod}-B}(M, [N, L]_C)$$
(2.37)

It implies

2.8. LEMMA. Let A, B be operads, $X \in \operatorname{Vect}(\mathbb{k})$ an A-B-bimodule. Then there is a canonical map of operads

$$\phi \colon A \to [X, X]_B \tag{2.38}$$

Indeed, by the adjunction above, maps $A \to [X, X]_B$ are in 1-to-1 correspondence with the maps of right *B*-modules $A \circ_A X \to X$. We know that $A \circ_A X = X$, so the map ϕ is the map corresponding to the identity map of right *B*-modules.

2.8.1. THE OPERADIC SHIFT. Let X be a symmetric sequence in $\mathcal{V}ect(\Bbbk)$. Define another symmetric sequence $X\{1\}$, called *the operadic shift* of X, as

$$X\{1\}(n) = X[-n+1] \otimes_{\Sigma_n} sgn_n \tag{2.39}$$

where sgn_n denotes the (1-dimensional) sign representation of σ_n .

We list the compatibility properties of the operadic shift with the monoidal structures and Hom's on $\mathcal{V}ect(\mathbb{k})_{\Sigma}$:

2.9. LEMMA. Let $X, Y \in \mathcal{V}ect(\mathbb{k})_{\Sigma}$. The following statements are true:

- $(i) \ (X \circ Y)\{1\} = X\{1\} \circ Y\{1\},$
- (*ii*) $[X, Y]{1} = [X{1}, Y{1}],$
- (*iii*) $(X \otimes_{lev} Y)\{1\} = (X\{1\}) \otimes_{lev} Y = X \otimes_{lev} (Y\{1\}),$
- (*iv*) Hom_{*lev*}(X{1}, Y{1}) = Hom_{*lev*}(X, Y),

2.10. COROLLARY. Let \mathcal{P} be an operad in $\operatorname{Vect}(\mathbb{k})$. Then $\mathcal{P}\{1\}$ is again an operad. Similarly, $\mathcal{C}\{1\}$ is a cooperad as soon as \mathcal{C} is a cooperad.

We denote

$$X\{n\} = (\dots ((X\{1\})\{1\})\dots)\{1\}$$
(2.40)

where the operation $-\{1\}$ is applied *n* times.

Let X be a vector space considered as the symmetric sequence $X^{(0)} = (X_{n=0}^{(0)}, 0, 0, ...)$. Then

$$X^{(0)}\{n\} = (X[n])^{(0)}$$
(2.41)

and

$$\operatorname{End}_{\mathfrak{O}p}(X)\{1\} = [X^{(0)}, X^{(0)}]\{1\} = \operatorname{End}_{\mathfrak{O}p}(X[1])$$
(2.42)

where -[n] is the conventional shift of degree of a vector space.

2.11. KOSZUL OPERADS. We refer the reader to [GK] and [LV, Ch. 7] for theory of Koszul operads.

Here we just fix some notations.

For a quadratic operad $\mathcal O$ we denote by $\mathcal O^!$ the quadratic dual cooperad. As well, we denote

$$\mathcal{O}^{i} = (\mathcal{O}\{1\})^{!} = \mathcal{O}^{!}\{-1\}$$
(2.43)

For any quadratic operad \mathcal{O} there is a map of dg operads

$$\operatorname{Bar}_{\mathcal{O}p}(\mathcal{O}^{\mathsf{i}}) \to \mathcal{O}$$
 (2.44)

inducing an isomorphism on H^0 . A quadratic operad \mathcal{O} is called *Koszul* if (2.44) is a quasi-isomorphism of dg operads. (See [LV, Section 6.5] for the definition of $\operatorname{Bar}_{\mathcal{O}p}(\mathcal{O})$).

Many classical operads are Koszul, among them Assoc, Comm, Lie, e_n . Their (shifted) Koszul dual are:

$$\begin{aligned} &\mathsf{Assoc}^{\mathsf{i}} = \mathsf{Assoc}^{*}\{-1\} \\ &\mathsf{Comm}^{\mathsf{i}} = \mathsf{Lie}^{*}\{-1\} \\ &\mathsf{Lie}^{\mathsf{i}} = \mathsf{Comm}^{*}\{-1\} \\ &\mathsf{e}_{d}^{\mathsf{i}} = \mathsf{e}_{d}^{*}\{-d\} \end{aligned} \tag{2.45}$$

2.12. Convolution complexes.

2.12.1. THE CONVOLUTION COMPLEX FOR (CO)OPERADS. Let \mathcal{P} be an operad, and \mathcal{C} a finite cooperad. Then $\operatorname{Hom}_{lev}(\mathcal{C}, \mathcal{P})$ is naturally an operad. Recall that its components are

$$\operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P})(n) = \operatorname{Hom}_{\Bbbk}(\mathcal{C}(n),\mathcal{P}(n))$$
(2.46)

and the symmetric group Σ_n acts on $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C},\mathcal{P})(n)$ as

$$(\sigma(f))(x) = \sigma(f(\sigma^{-1}x))$$

The operad structure is given as

$$\operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P}) \circ \operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P}) \to \operatorname{Hom}_{ev}(\mathcal{C} \circ_1 \mathcal{C}, \mathcal{P} \circ \mathcal{P}) \to \operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P})$$
(2.47)

Here the first arrow is (2.27), and the second arrow is given by the (co)operad structure maps $\mathcal{C} \to \mathcal{C} \circ_1 \mathcal{C}$ and $\mathcal{P} \circ \mathcal{P} \to \mathcal{P}$.

We denote this operad structure on $\operatorname{Hom}_{lev}(\mathcal{C}, \mathcal{P})$ by $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C}, \mathcal{P})$.

There is another construction, introduced in [KM], which associates a dg pre-Lie algebra \mathcal{P}_{Σ} with a dg operad \mathcal{P} .

The underlying complex of \mathcal{P}_{Σ} is defined as

$$\mathcal{P}_{\Sigma} = \bigoplus_{n \ge 1} \mathcal{P}(n)_{\Sigma_n} = \mathcal{P} \circ \mathbb{k}^{(0)}$$
(2.48)

(we assume that \mathcal{P} is a non-unital operad, that is, $\mathcal{P}(0) = 0$).

The formula for $\Psi_1 \circ \Psi_2 \in \mathcal{P}(m+n-1)$, for $\Psi_1 \in \mathcal{P}(m)$ and $\Psi_2 \in \mathcal{P}(n)$, reads:

$$\Psi_1 \star \Psi_2 = \sum_{i=1}^m \pm \Psi_1 \circ_i \Psi_2$$
 (2.49)

where

$$\Psi_1 \circ_i \Psi_2 = \Psi_1(\mathrm{id}^{\otimes (i-1)} \otimes \Psi_2 \otimes \mathrm{id}^{\otimes (m-i)})$$
(2.50)

The associated dg Lie algebra is also denoted by \mathcal{P}_{Σ} . For Ψ_1, Ψ_2 as above,

$$[\Psi_1, \Psi_2] = \Psi_1 \star \Psi_2 - (-1)^{|\Psi_1||\Psi_2|} \Psi_2 \star \Psi_1$$
(2.51)

See [MK, Sect. 1.7] for more detail.

We can apply the above construction to the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C}, \mathcal{P})$, see (2.47). We get the operadic convolution Lie algebra on the dg vector space

$$\operatorname{Conv}(\mathcal{C}, \mathcal{P}) = \operatorname{Hom}_{\mathfrak{O}p}(\mathcal{C}, \mathcal{P})_{\Sigma} = \prod_{k \ge 1} \operatorname{Hom}_{lev}(\mathcal{C}(k), \mathcal{P}(k))_{\Sigma_k}$$
(2.52)

2.13. REMARK. Assume an operad \mathcal{P} in $\mathcal{V}ect(\Bbbk)$ acts on a (dg) symmetric sequence $X \in \mathcal{V}ect(\Bbbk)_{\Sigma}$. Then the operad \mathcal{P} acts on the (dg) vector space X_{Σ} . Indeed, $X_{\Sigma} = X \circ \Bbbk^{(0)}$, and one has

$$\mathcal{P} \circ (X \circ \Bbbk^{(0)}) = (\mathcal{P} \circ X) \circ \Bbbk^{(0)} \xrightarrow{m_X \circ \mathrm{id}} X \circ \Bbbk^{(0)}$$

where $m_X \colon \mathcal{P} \circ X \to X$ defines the \mathcal{P} -action on X.

2.13.1. The convolution complex for (co)algebras. Let \mathcal{P} be an operad, \mathcal{C} a finite cooperad.

A coalgebra C over \mathcal{C} is given by its structure map $\Delta_C \colon C \to \mathcal{C} \hat{\circ}_1 C$. A coalgebra C is called *finite* if the map Δ_C factors as

$$C \to \mathcal{C} \circ_1 C \to \mathcal{C} \circ_1 C$$

Let $A \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ be an algebra over \mathcal{P} , and $C \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ a finite coalgebra over \mathcal{C} . Consider the symmetric sequence $\operatorname{Hom}_{lev}(C, A)$. We claim that it becomes an algebra over the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C}, \mathcal{P})$.

Indeed, there are maps

$$\operatorname{Hom}_{lev}(\mathcal{C},\mathcal{P}) \circ \operatorname{Hom}_{lev}(C,A) \to \operatorname{Hom}_{lev}(\mathcal{C} \circ_1 C, \mathcal{P} \circ A) \to \operatorname{Hom}_{lev}(C,A)$$
(2.53)

The first map is given by (2.27), and the second map is given by the compositions $C \to \mathcal{C} \circ_1 C$ and $\mathcal{P} \circ A \to A$.

There is a differential on $\operatorname{Hom}_{lev}(C, A)$, defined for a homogeneous Ψ as

$$(d\Psi)(x) = d_A(\Psi(x)) - (-1)^{|\Psi|} \Psi(d_C(x))$$
(2.54)

where d_A and d_C are the differentials in A and C, correspondingly.

When \mathcal{P} is a dg operad and \mathcal{C} is a dg cooperad, the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C}, \mathcal{P})$ is a dg operad, with the differential defined similarly to (2.54), via the differentials in \mathcal{C} and \mathcal{P} .

We get:

2.14. LEMMA. Let \mathcal{P} be an operad, \mathcal{C} a finite cooperad over $\operatorname{Vect}(\Bbbk)$. Let $A \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ be an algebra over \mathcal{P} , $C \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ a finite coalgebra over \mathcal{C} . Then the construction above makes the symmetric sequence $\operatorname{Hom}_{lev}(C, A)$ a dg algebra in $\operatorname{Vect}(\Bbbk)_{\Sigma}$ over the dg operad $\operatorname{Hom}_{\operatorname{Op}}(\mathcal{C}, \mathcal{P})$.

2.15. COROLLARY. In the notations as in Lemma 2.14, the dg operad $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{C}, \mathcal{P})$ acts on the dg vector space $\operatorname{Hom}_{\Sigma}(C, A)$

PROOF. It follows from Lemma 2.14 and Remark 2.13.

2.15.1. Let \mathcal{O} be a Koszul operad. Let C be a coalgebra over \mathcal{O}^i , and A an algebra over \mathcal{O} . Consider the symmetric sequence $\operatorname{Hom}_{lev}(C, A)$. By (2.53), it is an algebra over the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathcal{O}^i, \mathcal{O})$. The following property is very important for the deformation theory:

2.16. LEMMA. For any Koszul operad \mathcal{O} with finite-dimensional $\mathcal{O}(2)$, there is a map of operads

$$\phi: \operatorname{Lie}\{1\} \to \operatorname{Hom}_{\mathcal{O}p}(\mathcal{O}^{\mathsf{i}}, \mathcal{O}) \tag{2.55}$$

The map ϕ was constructed in [GK] via the operadic analogues of Manin's black and white products of quadratic algebras, see [GK, 2.2], [LV, 8.8].

Later we use the following explicit formula for the arity component $\phi(2)$.

Let $E = \mathcal{O}(2)$, $E[1] = \mathcal{O}^{i}(2)$. Take any basis t_1, \ldots, t_m in E. Denote by $t_1[1], \ldots, t_m[1]$ the corresponding basis in E[1]. Then the map ϕ sends the canonical generator $\omega = [-, -]$ in Lie $\{1\}(2)$ to

$$\phi(\omega) = \sum_{i=1}^{m} t_i[1]^* \otimes t_i \in \operatorname{Hom}_{\Sigma_2}(\mathcal{O}^{\mathsf{i}}(2), \mathcal{O}(2))$$
(2.56)

As the operad Lie is quadratic, the component $\phi(2)$ defines the map ϕ .

2.17. COROLLARY. In the assumptions as above, the symmetric sequence

$$\operatorname{Hom}_{lev}(C,A)\{-1\}$$

has a natural structure of a Lie algebra.

The symmetric sequence $\operatorname{Hom}_{lev}(C, A)\{-1\}$ with its Lie algebra structure is often called *the convolution symmetric sequence*.

Consider an element $f \in \prod_{n \ge 1} \operatorname{Hom}_{lev}(C, A) \{-1\}(n)^1$ satisfying the Maurer-Cartan equation:

$$d_0 f + \frac{1}{2} [f, f] = 0 \tag{2.57}$$

In this situation, one can twist the differential in $\operatorname{Hom}_{lev}(C, A)\{-1\}$ by $\operatorname{ad}(f)$, defined as $\operatorname{ad}(f)(-) := [f, -]$, where [-, -] is the convolution Lie bracket, and get a dg Lie algebra

$$(\operatorname{Hom}_{\Sigma}(C, A)\{-1\}, d_0 + \operatorname{ad}(f))$$
 (2.58)

3. The bar-cobar adjunction and its relative version

3.1. PRO-CONILPOTENT \mathcal{O} -COALGEBRAS. Let \mathcal{O} be an operad, $V \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ a symmetric sequence. Denote by $\mathcal{A}lg(\mathcal{O})$ the category of non-unital \mathcal{O} -algebras in $\mathcal{V}ect(\Bbbk)_{\Sigma}$.

The forgetful functor $\mathcal{A}lg(\mathcal{O}) \to \mathcal{V}ect(\Bbbk)_{\Sigma}$, from \mathcal{O} -algebras in $\mathcal{V}ect(\Bbbk)_{\Sigma}$ to symmetric sequences, admits a left adjoint. It is given by the free algebra over \mathcal{O} , generated by V:

$$\mathcal{F}_{\mathcal{O}}(V) = \mathcal{O} \circ V \tag{3.1}$$

$$\operatorname{Hom}_{\mathcal{A}lg(\mathcal{O})}(\mathcal{F}_{\mathcal{O}}(V), A) = \operatorname{Hom}_{\operatorname{Vect}(\Bbbk)_{\Sigma}}(V, A)$$
(3.2)

It is a non-unital O-algebra.

There is a version of it for the unital augmented \mathcal{O} -algebras. Denote by $\mathcal{A}lg^{u,aug}(\mathcal{O})$ the category of such algebras over \Bbbk . For $A \in \mathcal{A}lg^{u,aug}(\mathcal{O})$, define the functor

$$R(A) = A_{+} = \operatorname{Ker}(\varepsilon \colon A \to \mathbb{k}^{(0)})$$
(3.3)

where $\varepsilon \colon A \to \mathbb{k}^{(0)}$ is the augmentation, which is assumed to be a map of \mathcal{O} -algebras. The functor R admits a left adjoint:

$$\mathcal{F}^{u}_{\mathcal{O}}(V) = (\mathcal{O} \circ V) \oplus \mathbb{k}^{(0)}$$
(3.4)

One has:

$$\operatorname{Hom}_{\mathcal{A}lg^{u,aug}(\mathcal{O})}(\mathcal{F}^{u}_{\mathcal{O}}(V),A) = \operatorname{Hom}_{\operatorname{Vect}(\Bbbk)_{\Sigma}}(V,A_{+})$$
(3.5)

Such a (right) adjoint functor to the forgetful functor does not exist, in general, for the coalgebras over an operad. One should restrict ourselves to a class of coalgebras, called *pro-conilpotent*. (In [T2], they are called pro-coartinian. In [Q, Appendix B] they are

called connected.) In this paper, we consider pro-conilpotent C-coalgebras in symmetric sequences, where C is a finite cooperad, see (2.10).

For simplicity, assume that the finite cooperad C is *quadratic* (it is the only case we deal in this paper with). Assume that C is an *biaugmented* cooperad, where by a *biaugmentation* we mean operad maps $I \xrightarrow{\eta} C \xrightarrow{\varepsilon} I$.

Let \mathcal{C} be a quadratic biaugmented cooperad. We denote

$$\mathcal{C}(n)_+ = \operatorname{Ker}(\varepsilon \colon \mathcal{C}(n) \to I(n))$$

By a \mathcal{C} -coalgebra we understand an object $C \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ with a structure map $\Delta : C \to \mathcal{C} \hat{\circ}_1 C$ and with a counit $\varepsilon_1 : C \to \Bbbk^{(0)}$, satisfying the natural axioms. By a coaugmentation of the coalgebra C we mean a map $\eta_1 : \Bbbk^{(0)} \to C$ of \mathcal{C} -coalgebras, such that $\varepsilon_1 \circ \eta_1 = \mathrm{id}$.

Set $F^0C = \text{Im}(\eta_1) \simeq \mathbb{k}^{(0)}$. The symmetric sequence C/F^0C gets a structure of a non-counital C-coalgebra. Define an ascending filtration on C/F^0C , as follows.

Set

$$F^{i}(C/F^{0}C) = \left\{ x \in C/F^{0}C | \Delta(x) = (\underset{1}{0}, \dots, \underset{i}{0}, t_{i+1}, t_{i+2}, \dots \right\}$$

where $t_{\ell} \in (\mathcal{C}(\ell) \otimes C^{\otimes \ell})^{\Sigma_{\ell}}$.

One has

$$0 \subset F^1(C/F^0C) \subset F^2(C/F^0C) \subset F^3(C/F^0C) \subset \dots$$

A coaugmented C-coalgebra C is called *pro-conilponent* if the ascending filtration $\{F^i(C/F^0C)\}_{i>1}$ is *exhausted*.

Denote the category of pro-conilpotent counital coaugmented coalgebras over a cooperad \mathcal{C} by $\mathcal{C}oalg_{\text{pronilp}}^{cu,caug}(\mathcal{C})$.

Consider the forgetful functor $\mathcal{V}: \operatorname{Coalg}_{\operatorname{pronilp}}^{cu,caug}(\mathcal{C}) \to \operatorname{Vect}(\Bbbk)_{\Sigma}$, defined as

$$\mathcal{V}(C) = C/F^0 C \tag{3.6}$$

One has:

3.2. LEMMA. Let C be a finite quadratic biaugmented cooperad. Then the functor

$$\mathcal{V}\colon \mathfrak{C}oalg_{\mathrm{pronilp}}^{cu,caug}(\mathcal{C}) \to \mathcal{V}ect(\Bbbk)_{\Sigma}$$

admits a right adjoint F, given by the direct sum cofree coalgebra

$$\mathcal{F}^{\mathcal{C}}(V) = \left(\bigoplus_{n \ge 1} (\mathcal{C}(n) \otimes V^{\boxtimes n})^{\Sigma_n}\right) \oplus \mathbb{k}^{(0)}$$
(3.7)

See e.g. [T2, Prop. 2.3] for a proof.

We refer to $\mathcal{F}^{\mathcal{C}}(V)$ as the cofree *C*-coalgebra cogenerated by *V*. One also has:

3.3. LEMMA. Let C be a finite quadratic biaugmented cooperad, C a pro-conilpotent coalgebra over C. Then the structure map

$$\Delta \colon C/F^0C \to \mathcal{C}\hat{\circ}_1(C/F^0C)$$

factors as

$$C/F^0C \to \mathcal{C} \circ_1 (C/F^0C) \to \mathcal{C} \circ_1 (C/F^0C)$$

where the second map is the canonical imbedding.

See [T2, Lemma 2.2] for a proof.

3.4. THE BAR DG SYMMETRIC SEQUENCE. Let $X \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ be a symmetric sequence, \mathcal{O} a biaugmented Koszul operad with finite-dimensional components $\mathcal{O}(n)$. Denote by \mathcal{O}^{i} the shifted Koszul dual cooperad, see (2.43).

Let $X \in \operatorname{Vect}(\mathbb{k})_{\Sigma}$ be an \mathcal{O} -algebra. Then the cofree \mathcal{O}^{i} -coalgebra $\mathcal{F}^{\mathcal{O}^{i}}(V)$ is endowed with a component-wise differential, as follows.

Recall the twisting morphism $\kappa \colon \mathcal{O}^i \to \mathcal{O}$ of degree 1, see Lemma 2.16. One has the following composition

$$\mathcal{O}^{\mathsf{i}} \circ_1 X \xrightarrow{\operatorname{can}_{\mathcal{O}^{\mathsf{i}}, X}} \mathcal{O}^{\mathsf{i}} \circ X \xrightarrow{\operatorname{\kappaoid}} \mathcal{O} \circ X \xrightarrow{m_X} X \tag{3.8}$$

see (2.14) for the map *can*.

It is a map of symmetric sequences of degree +1. It can be extended to a coderivation of $\mathcal{O}^{i} \circ X$ as of \mathcal{O}^{i} -coalgebra, see [LV, 11.2.2]. All maps used in this extension are maps of symmetric sequences.

One denotes d_{Bar} the corresponding map. One has

$$d_{\text{Bar}}^2 = 0$$

We denote by $\operatorname{Bar}_{\mathcal{O}}(X)$ the cofree coalgebra $\mathcal{F}_{\mathcal{O}^{i}}(X)$ over \mathcal{O}^{i} endowed with this differential. The differential agrees with the cooperations by the Leibniz rule, and acts component-wise. As well, the differential commutes with the action of symmetric group(s).

It is a pro-conilpotent coalgebra over \mathcal{O}^{i} in $\mathcal{V}ect(\mathbb{k})_{\Sigma}$.

When X is an arity 0 symmetric sequence, $\operatorname{Bar}_{\mathcal{O}}(X)$ is an arity 0 symmetric sequence as well, and agrees with the conventional definition.

3.5. THE COBAR DG SYMMETRIC SEQUENCE. Let Y be a pro-conilpotent \mathcal{O}^{i} -coalgebra in $\operatorname{Vect}(\mathbb{k})_{\Sigma}$. One defines its cobar dg symmetric sequence as

$$\operatorname{Cobar}_{\mathcal{O}^{i}}(Y) = (\mathcal{F}_{\mathcal{O}}(Y/F^{0}Y), \ d_{\operatorname{Cobar}}) = (\mathcal{O} \circ (Y/F^{0}Y), d_{\operatorname{Cobar}})$$
(3.9)

where the cobar-differential d_{Cobar} is defined as follows.

The symmetric sequence $\mathcal{O} \circ (Y/F^0Y)$ is an \mathcal{O} -algebra. The differential is defined to satisfy the Leibniz rule, so it is defined by its restriction to the generators. Consider a map of symmetric sequences of degree +1

$$d_{\text{Cobar}} \colon Y/F^0 Y \to \mathcal{O} \circ (Y/F^0 Y) \tag{3.10}$$

defined as

$$Y/F^{0}Y \to \mathcal{O}^{\mathsf{i}} \circ_{1} (Y/F^{0}Y) \xrightarrow{\operatorname{can}_{\mathcal{O}^{\mathsf{i}}, Y/F^{0}Y}} \mathcal{O}^{\mathsf{i}} \circ (Y/F^{0}Y) \xrightarrow{\kappa \circ \mathrm{id}} \mathcal{O} \circ (Y/F^{0}Y) \tag{3.11}$$

Here the first map is given by the \mathcal{O}^{i} -coalgebra structure on $Y/F^{0}Y$ (inherited by the \mathcal{O}^{i} -coalgebra structure on Y).

One has

 $d_{\rm Cobar}^2 = 0$

As for the case of the bar-differential, the extension of (3.11) by Leibniz rule as an \mathcal{O} -algebra derivation, uses only maps of symmetric sequences. Therefore, the differential d_{Cobar} acts component-wise.

3.6. THE BAR-COBAR ADJUNCTION. Let \mathcal{O} be a Koszul operad in $\mathcal{V}ect(\Bbbk)$, with finitedimensional $\mathcal{O}(2)$, such that the cooperad \mathcal{O}^{i} is bi-augmented, X an \mathcal{O} -algebra, Y a pro-conilpotent \mathcal{O}^{i} -coalgebra. Consider the dg vector space $\operatorname{Hom}_{\Sigma}(Y/F^{0}Y, X)$. The symmetric sequence $Y/F^{0}Y$ is a coalgebra over \mathcal{O}^{i} . Then it follows from Lemma 2.14 that $\operatorname{Hom}_{\Sigma}(Y/F^{0}Y, X)$ is an algebra over the operad $\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O})$. By Lemma 2.16, there is a map of operads $\operatorname{Lie}\{1\} \to \operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O})$. Therefore, by Remark 2.13 (or by Corollary 2.15), the dg vector space $\operatorname{Hom}_{\Sigma}(Y/F^{0}Y, X)[-1]$ gets a (dg) Lie algebra structure.

3.7. PROPOSITION. Let \mathcal{O} be a Koszul operad with finite-dimensional components $\mathcal{O}(n)$, X an \mathcal{O} -algebra, Y a pro-conilpotent \mathcal{O}^{i} -coalgebra. Assume that the cooperad \mathcal{O}^{i} is finite and biaugmented. One has the following functorial isomorphisms of sets:

 $\operatorname{Hom}_{\Sigma,\mathcal{A}lg(\mathcal{O})}(\operatorname{Cobar}_{\mathcal{O}^{\mathsf{i}}}(Y),X) = \operatorname{MC}(\operatorname{Hom}_{\Sigma}(Y/F^{0}Y,X)[-1]) = \operatorname{Hom}_{\Sigma,\operatorname{Coalg}(\mathcal{O}^{\mathsf{i}})}(Y,\operatorname{Bar}_{\mathcal{O}}(X))$ (3.12)

where $\operatorname{Coalg}(\mathcal{O}^{i})$ stands for the category of pro-conlipotent \mathcal{O}^{i} -coalgebras.

The set in the middle of (3.12) is called the set of *twisted morphisms*.

PROOF. The proof repeats the well-known argument for the case when X and Y are dg vector spaces (considered as arity-zero symmetric sequences), see e.g. [LV, 11.3.1]. We make use the adjunction given by Lemma 3.2 (as well as its more straightforward counter-part for $\mathcal{A}lg(\mathcal{O})$). The compatibility with the (co)bar-differentials is translated to the Maurer-Cartan equation in $\operatorname{Hom}_{\Sigma}(Y/(F^0Y), X)[-1]$.

3.8. A FRAGMENT OF THE \mathcal{P} -RELATIVE BAR-COBAR ADJUNCTION. Recall that a symmetric sequence Z is a right \mathcal{P} -module, where \mathcal{P} is an operad, if there is a map

$$m_Z \colon Z \circ \mathcal{P} \to Z$$

satisfying the natural associativity and unit axioms.

Let Z be a pro-conilpotent coalgebra over a cooperad \mathcal{C} and a right module over an operad \mathcal{P} .

We say that these two structures are compatible, if the following conditions are fulfilled:

(i) the right \mathcal{P} -module structure on Z descents to a right \mathcal{P} -module structure on $Z/(F^0Z)$, that is

$$m_Z(F^0(Z) \circ \mathcal{P}) \subset F^0(Z) \tag{3.13}$$

(ii)

$$\begin{array}{ccc} (Z/F^{0}Z) \circ \mathcal{P} & \xrightarrow{m_{\mathcal{P}}} Z/F^{0}Z & \xrightarrow{\Delta_{Z}} \mathcal{C} \circ_{1} (Z/F^{0}Z) \xrightarrow{can} \mathcal{C} \circ (Z/F^{0}Z) \\ & \downarrow = \\ (\mathcal{C} \circ_{1} (Z/F^{0}Z)) \circ \mathcal{P} \xrightarrow{can} (\mathcal{C} \circ (Z/F^{0}Z)) \circ \mathcal{P} \to \mathcal{C} \circ ((Z/F^{0}Z) \circ \mathcal{P}) \to \mathcal{C} \circ (Z/F^{0}Z) \\ & (3.14) \end{array}$$

Here can denotes the canonical isomorphism (2.13).

We denote by $\mathcal{C}(\mathcal{C}, \mathcal{P})$ the category of symmetric sequences Z with the above conditions.

For any right \mathcal{P} -module Z and any cooperad \mathcal{C} , the cofree coalgebra $\mathcal{F}^{\mathcal{C}}(Z)$ is naturally an object of $\mathfrak{C}(\mathcal{C}, \mathcal{P})$.

One has:

3.9. LEMMA. Let \mathcal{O} be a Koszul operad, \mathcal{P} an operad, X a $\mathcal{O}-\mathcal{P}$ -bimodule. Then the bar dg symmetric sequence $\operatorname{Bar}_{\mathcal{O}}(X)$ is a dg object of the category $\mathcal{C}(\mathcal{O}^{i}, \mathcal{P})$, what amounts to say that the bar-differential is compatible with the right \mathcal{P} -action.

For two right \mathcal{P} -modules $M, N \in \mathcal{V}ect(\mathbb{k})_{\Sigma}$, define $\operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(M,N) \in \mathcal{V}ect(\mathbb{k})_{\Sigma}$ as the equalizer

$$\operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(M,N) \to \operatorname{Hom}_{lev}(M,N) \stackrel{u}{\xrightarrow{}}_{v} \operatorname{Hom}_{lev}(M \circ \mathcal{P},N)$$
(3.15)

where, for $f \in \operatorname{Hom}_{lev}(M, N)$, u(f) is defined as the composition $M \circ \mathcal{P} \xrightarrow{\mu_M} M \xrightarrow{f} N$, and v(f) is defined as the composition $M \circ \mathcal{P} \xrightarrow{f \circ \operatorname{id}} N \circ \mathcal{P} \xrightarrow{\mu_N} N$.

One easily shows that

$$\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(M,N) = \operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(M,N) \circ \mathbb{k}^{(0)}$$
(3.16)

(the functor of invariants is right adjoint and thus commutes with the limits, and in the case of char $\mathbf{k} = 0$ the spaces of invariants and coinvariants are canonically isomorphic).

Let \mathcal{O} be a Koszul operad, \mathcal{P} an operad, X an \mathcal{O} - \mathcal{P} -bimodule, $Y \in \mathcal{C}(\mathcal{O}^{i}, \mathcal{P})$.

Below we construct a dg Lie algebra structure on $\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(Y/\Bbbk, X)[-1]$. (For the case $\mathcal{P} = I$, this Lie algebra coincides with the one which figures in the middle term of (3.12)).

To this end, we construct an operad action

$$\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev, \operatorname{Mod}-\mathcal{P}}(Y/F^{0}Y, X) \to \operatorname{Hom}_{lev, \operatorname{Mod}-\mathcal{P}}(Y/F^{0}Y, X)$$

For a general functor $F \colon \mathfrak{C} \to \mathfrak{D}$, and a diagram $D \colon I \to \mathfrak{C}$, there is a canonical morphism in \mathfrak{D} :

$$\xi \colon F(\lim D(i)) \to \lim F(D_i) \tag{3.17}$$

Therefore, one has a morphism in $\operatorname{Vect}(\Bbbk)_{\Sigma}$:

$$\xi \colon \operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev, \operatorname{Mod}-\mathcal{P}}(Y/F^{0}Y, X) \to \lim \left(\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev}(Y/F^{0}Y, X) \stackrel{\operatorname{id} \circ u}{\underset{\operatorname{id} \circ v}{\rightrightarrows}} \operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev}((Y/F^{0}Y) \circ \mathcal{P}, X) \right)$$

$$(3.18)$$

The limit in the r.h.s. of (3.18) is further mapped to

$$\lim \left(\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev}(Y/F^{0}Y, X) \stackrel{\operatorname{id} \circ u}{\rightrightarrows} \operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev}((Y/F^{0}Y) \circ \mathcal{P}, X) \right) \\ \stackrel{(2.27)}{\rightarrow} \lim \left(\operatorname{Hom}_{lev}(\mathcal{O}^{i} \circ (Y/F^{0}Y), \mathcal{O} \circ X) \stackrel{u_{1}}{\rightrightarrows} \operatorname{Hom}_{lev}(\mathcal{O}^{i} \circ ((Y/F^{0}Y) \circ \mathcal{P}), \mathcal{O} \circ X) \right) \\ = \lim \left(\operatorname{Hom}_{lev}(\mathcal{O}^{i} \circ (Y/F^{0}Y), \mathcal{O} \circ X) \stackrel{u_{2}}{\rightrightarrows} \operatorname{Hom}_{lev}((\mathcal{O}^{i} \circ (Y/F^{0}Y)) \circ \mathcal{P}), \mathcal{O} \circ X) \right) \stackrel{*}{\rightarrow} \\ \lim \left(\operatorname{Hom}_{lev}(Y/F^{0}Y, X) \stackrel{u}{\Longrightarrow} \operatorname{Hom}_{lev}((Y/F^{0}Y) \circ \mathcal{P}, X) \right) \\ = \operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(Y/F_{0}Y, X) \xrightarrow{u} \left(\operatorname{Hom}_{lev}(Y/F^{0}Y, X) \stackrel{u}{\Longrightarrow} \right) \\ (3.19)$$

(the arrows u_1, v_1 and u_2, v_2 are clear, and we skip their definitions). Here the crucial point is the arrow marked by * in the third line. It encodes both diagram (3.14) (expressing that $Y \in \mathcal{C}(\mathcal{O}^i, \mathcal{P})$), and the property of X being an $\mathcal{O}-\mathcal{P}$ -bimodule. Indeed, the equality follows from commutativity of the two diagrams, corresponded to the cases $\omega = u$ and $\omega = v$ of the diagram below:

The commutativity of (3.20) follows from (3.14), whereas its commutativity for $\omega = v$ follows from the fact that X is an $\mathcal{O}-\mathcal{P}$ -bimodule.

One easily shows that the constructed map of symmetric sequences

$$\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O}) \circ \operatorname{Hom}_{lev, \operatorname{Mod}-\mathcal{P}}(Y/F^{0}Y, X) \to \operatorname{Hom}_{lev, \operatorname{Mod}-\mathcal{P}}(Y/F^{0}Y, X)$$

gives rise to an action of the operad $\operatorname{Hom}_{lev}(\mathcal{O}^{i}, \mathcal{O})$ (see Corollary 2.7) on

$$\operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(Y/F^0Y,X) \in \operatorname{Vect}(\Bbbk)_{\Sigma}$$

One has a map of operads $\text{Lie}\{1\} \to \text{Hom}_{\mathcal{O}p}(\mathcal{O}^{i}, \mathcal{O})$, by Lemma 2.16. Therefore, $\text{Hom}_{lev,\text{Mod}-\mathcal{P}}(Y/F^{0}Y, X)$ becomes a $\text{Lie}\{1\}$ -algebra. The same is true for

 $\operatorname{Hom}_{lev,\operatorname{Mod}-\mathcal{P}}(Y/F^0Y,X) \circ \mathbb{k}$, see Remark 2.13. The latter space is identified with $\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(Y/F^0Y,X)$, see (3.16).

It is an algebra over Lie{1}. Therefore, $\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{P}}(Y/F^0Y,X)[-1]$ is a dg Lie algebra (the differential is d_0 , that is, it comes from the inner differentials on X and Y).

Now we can formulate our statement:

3.10. PROPOSITION. Let \mathcal{O} be a Koszul biaugmented operad, \mathcal{P} an operad, $X \in \mathcal{O}-\mathcal{P}$ bimodule, Y an object of the category $\mathcal{C}(\mathcal{O}^{i}, \mathcal{P})$. One has:

$$\mathrm{MC}(\mathrm{Hom}_{\Sigma,\mathrm{Mod}-\mathcal{P}}(Y/(F^{0}Y),X)[-1]) = \mathrm{Hom}_{\Sigma,\mathfrak{C}(\mathcal{O}^{\mathrm{i}},\mathcal{P})}(Y,\mathrm{Bar}_{\mathcal{O}}(X))$$
(3.21)

PROOF. After all preparations, the statement becomes almost trivial. A map $t: Y \to \text{Bar}_{\mathcal{O}}(X)$ of the underlying coalgebras is the same that a map $t': Y/F^0Y \to X$, by Lemma 3.2. The map t is a map of right \mathcal{P} -modules if and only if the map t' is. Finally, the compatibility of t with the bar-differential results in the Maurer-Cartan equation on t'.

- 4. Deformation theory of a morphism of operads $\mathbf{e}_n \to \mathcal{P}$ with a cocommutative base
- 4.1. The case $\mathcal{P} = [X, X]$.

4.1.1. Let $V \in \operatorname{Vect}(\Bbbk)$ be a (dg) vector space over \Bbbk , $X \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ a symmetric sequence. Recall the symmetric sequence $V \star X := V^{(0)} \boxtimes X$ (that is, $(V \star X)(n) = V \otimes X(n)$).

One has the adjunction

$$\operatorname{Hom}_{\Sigma}(V \star X, Y) = \operatorname{Hom}_{\Bbbk}(V, \operatorname{Hom}_{\Sigma}(X, Y))$$

$$(4.1)$$

One has the following fact:

4.2. LEMMA. Let $V \in \operatorname{Vect}(\mathbb{k})$, $X \in \operatorname{Vect}(\mathbb{k})_{\Sigma}$. The following statements are true:

- (i) Let $\mathcal{P}_1, \mathcal{P}_2$ are operads, V is an algebra over \mathcal{P}_1, X an algebra over \mathcal{P}_2 . Then $V \star X$ is an algebra over the operad $\mathcal{P}_1 \otimes_{lev} \mathcal{P}_2$.
- (ii) Let C_1, C_2 are cooperads, V is a coalgebra over C_1, X is a coalgebra C_2 . Then $V \star X$ is a coalgebra over $C_1 \otimes_{lev} C_2$.

Proof.

(i): One has maps

$$\mathcal{P}_1(k) \otimes_{\Sigma_k} V^{\otimes k} \to V$$

and

$$\mathcal{P}_2(k) \otimes_{\Sigma_k} \operatorname{Ind}_{\Sigma_{n_1} \times \cdots \times \Sigma_{n_k}}^{\Sigma_{n_1} + \cdots + n_k} (X(n_1) \otimes \cdots \otimes X(n_k)) \to X(n_1 + \cdots + n_k)$$

They give

$$(\mathcal{P}_1(k) \otimes_{lev} \mathcal{P}_2(k)) \otimes_{\Sigma_k} \operatorname{Ind}_{\Sigma_{n_1} \times \dots \times \Sigma_{n_k}}^{\Sigma_{n_1} + \dots + n_k} (V^{\otimes k} \otimes X(n_1) \otimes \dots \otimes X(n_k)) \to V \otimes X_{n_1 + \dots + n_k}$$

which gives a structure of an algebra over $\mathcal{P}_1 \otimes_{lev} \mathcal{P}_2$ on $V \star X$.

(ii): is analogous.

4.2.1. Let \mathcal{O} be a Koszul operad, $X \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ an \mathcal{O} -algebra. Recall the dg sequence

$$\operatorname{Bar}_{\mathcal{O}}(X) = (\mathcal{O}^{\mathsf{i}} \circ X, d_0 + d_{\operatorname{Bar}})$$

see Section 3.4. It is a coalgebra over the cooperad \mathcal{O}^{i} .

Let $X, Y \in \mathcal{V}ect(\Bbbk)_{\Sigma}$ be \mathcal{O} -algebras, $f: X \to Y$ a map of \mathcal{O} -algebra. It defines a map

$$\operatorname{Bar}(f) \colon \operatorname{Bar}_{\mathcal{O}}(X) \to \operatorname{Bar}_{\mathcal{O}}(Y)$$

of dg symmetric sequences of coalgebras over \mathcal{O}^{i} .

We prefer to deal with coalgebras over finite biaugmented cooperads, as the concept of a pro-conilpotent coalgebra over a cooperad, see Section 3.1, necessary for the adjunction in Proposition 3.7, exists only for such cooperads. In our case, $\mathcal{O} = \mathbf{e}_n$, the cooperad \mathbf{e}_n^i is not biaugmented. On the other hand, $\mathbf{e}_n^i = \mathbf{e}_n^* \{-n\}$, and the cooperad \mathbf{e}_n^* is biaugmented. Therefore, we replace the map $\operatorname{Bar}(f)$ of $\mathbf{e}_n^!$ -coalgebras by the corresponding map of $\mathbf{e}_n^i \{n\} = \mathbf{e}_n^*$ -coalgebras.

Consider

$$\operatorname{Bar}(f)\{n\}\colon\operatorname{Bar}_{\mathsf{e}_n^{\mathsf{i}}}(X)\{n\}\to\operatorname{Bar}_{\mathsf{e}_n^{\mathsf{i}}}(Y)\{n\}$$

One has:

$$(\mathbf{e}_{n}^{i} \circ X)\{n\} = (\mathbf{e}_{n}^{i}\{n\}) \circ (X\{n\}) = \mathbf{e}_{n}^{*} \circ (X\{n\})$$
(4.2)

Denote

$$Bar_n(X) = (\mathbf{e}_n^* \circ (X\{n\}), d_{Bar}\{n\})$$
(4.3)

By abuse of notations, we will use the notation d_{Bar} for both the shifted differential $d_{\text{Bar}}\{n\}$ and the original bar differential.

Let a be a coalgebra over Comm^{*}. By Lemma 4.2,

$$a \star \operatorname{Bar}_n(X)$$

is a coalgebra over $\mathsf{Comm}^* \otimes \mathsf{e}_n^* = \mathsf{e}_n^*$.

Let $f: X \to Y$ be as above. Denote by

$$Coalg = Coalg(Comm^*)$$

the category of pro-conilpotent cocommutative coalgebras over k. Define the functor

$$F_{X,Y}^f \colon \mathfrak{Coalg} \to \mathfrak{Sets}$$

, as follow:

$$F_{X,Y}^f(a) = \left\{ \phi \in \operatorname{Hom}_{\Sigma,\operatorname{Coalg}(\mathbf{e}_n^*)} \left(a \star \operatorname{Bar}_n(X), \operatorname{Bar}_n(Y) \right), \quad \phi \circ \eta = \operatorname{Bar}(f)\{n\} \right\}$$
(4.4)

Here $\eta \colon \mathbb{k} \to a$ is the coaugmentation of a.

Our first task is to show that this functor is representable.

4.2.2. The following Proposition easily follows from the bar-cobar duality in symmetric sequences, proven in Section 3.6.

4.3. PROPOSITION. Let $X \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ be a e_n -algebra, and let $C \in \operatorname{Vect}(\Bbbk)_{\Sigma}$ be a proconlipotent e_n^* -coalgebra. One has:

$$\operatorname{Hom}_{\Sigma,\operatorname{Coalg}(\mathbf{e}_n^*)}(C,\operatorname{Bar}_n(X)) = \operatorname{MC}\left(\operatorname{Hom}_{\Sigma}(C/F^0C,X\{n\})[-1]\right)$$
(4.5)

where $\operatorname{Coalg}(\mathbf{e}_n^*)$ stands for the category of pro-conilpotent \mathbf{e}_n^* -coalgebras.

PROOF. It follows from Proposition 3.7 for $\mathcal{O} = \mathbf{e}_n\{n\}$.

_

4.3.1. By Proposition 4.3, one has:

$$F_{X,Y}^f(a) = \left\{ \theta \in \mathrm{MC}\big(\mathrm{Hom}_{\Sigma}((a \star \mathrm{Bar}_n(X))/\Bbbk, Y\{n\})[-1]\big), \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}} \right\}$$
(4.6)

where

$$\operatorname{Bar}(f)_{\operatorname{pr}} \in \operatorname{MC}(\operatorname{Hom}_{\Sigma}(\operatorname{Bar}_{n}(X\{n\})/\Bbbk, Y\{n\})[-1])$$

$$(4.7)$$

is the element corresponded to Bar(f) by (4.5).

One further has:

$$F_{X,Y}^f(a) = \left\{ \theta \in \mathrm{MC}\big(\mathrm{Hom}_{\mathbb{k}}(a, \mathrm{Hom}_{\Sigma}(\mathrm{Bar}_n(X\{n\})/\mathbb{k}, Y\{n\})[-1])\big), \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}} \right\}$$
(4.8)

We made use the adjunction (4.1).

Consider the graded vector space

$$\operatorname{Hom}_{\Sigma}(\operatorname{Bar}_{n}(X)/\Bbbk, Y\{n\}) = \operatorname{Hom}_{\Sigma}(\mathsf{e}_{n}^{*} \circ X\{n\}, Y\{n\})$$

By Lemma 2.14, it is an algebra over the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathbf{e}_n^*, \mathbf{e}_n\{n\}) = \operatorname{Hom}_{\mathcal{O}p}(\mathbf{e}_n^i, \mathbf{e}_n).$

By Lemma 2.16, there is a map of operads

$$\operatorname{Lie}\{1\} \to \operatorname{Hom}_{\mathcal{O}p}(\mathcal{O}^{\mathsf{i}}, \mathcal{O})$$

$$(4.9)$$

for any Koszul operad \mathcal{O} .

It makes $\operatorname{Hom}_{\Sigma}(\mathcal{O}^{i} \circ X, Y)[-1]$ is an algebra over the same operad Lie. In our situation, it implies that $\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*} \circ X\{n\}, Y\{n\})[-1]$ is a graded Lie algebra. One easily sees that the inner differential d_{0} (which is equal to 0 provided the differentials on X, Y are 0), and the differential d_{Bar} , are compatible with this Lie algebra structure.

Consider the dg Lie algebra

$$Def_{0}(X,Y) = (Hom_{\Sigma}(Bar_{n}(X)/\Bbbk, Y\{n\})[-1], d = d_{0} + d_{Bar}) = (Hom_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}])[-1], d = d_{0} + d_{Bar}^{\sim})$$
(4.10)

where d_{Bar}^{\sim} is the differential corresponded to d_{Bar} by the adjunction, and $[X\{n\}, Y\{n\}] = [X, Y]\{n\}$ by Lemma 2.9.

ON A HIGHER STRUCTURE ON OPERADIC DEFORMATION COMPLEXES 1013

The element $\operatorname{Bar}(f)_{\operatorname{pr}}$ is a degree 1 Maurer-Cartan element in $\operatorname{Def}_0(X,Y)$. Denote

$$d_f = \operatorname{ad}(\operatorname{Bar}(f)_{\operatorname{pr}}) \tag{4.11}$$

and define

$$\operatorname{Def}(X \xrightarrow{f} Y) = \left(\operatorname{Hom}_{\Sigma}(\operatorname{Bar}_{n}(X)/\mathbb{k}, Y\{n\})[-1], d = d_{0} + d_{\operatorname{Bar}} + d_{f}\right)$$

= $\left(\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X, Y]\{n\})[-1], d = d_{0} + d_{\operatorname{Bar}}^{\sim} + d_{f}^{\sim}\right)$ (4.12)

It is a dg Lie algebra.

4.3.2. One has:

4.4. PROPOSITION. Let X, Y be \mathbf{e}_n -algebras in $\operatorname{Vect}(\mathbb{k})_{\Sigma}$, and let $f: X \to Y$ be a map of \mathbf{e}_n -algebras. The functor $F_{X,Y}^f$ is representable, by the dg coalgebra

$$a_{\rm rep} = C_{\rm CE}({\rm Def}(X \xrightarrow{f} Y), \Bbbk)$$

$$(4.13)$$

which is the Chevalley-Eilenberg chain complex of the dg Lie algebra $\text{Def}(X \xrightarrow{f} Y)$. PROOF. (Cf. [T2, Prop. 3.2]). By (4.8), one has

$$F_{X,Y}^{f}(a) = \left\{ \theta \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a, \mathrm{Hom}_{\Sigma}(\mathrm{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1])), \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}} \right\}$$
(4.14)

Denote

$$L_{0} = (\operatorname{Hom}_{\Sigma}(\operatorname{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1], \ d = d_{0} + d_{\operatorname{Bar}})$$

$$L_{a} = (\operatorname{Hom}_{\Bbbk}(a, \operatorname{Hom}_{\Sigma}(\operatorname{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1]), \ d = d_{0} + d_{a} + d_{\operatorname{Bar}})$$
(4.15)

where d_a is the component coming from the differential on the dg coalgebra a.

There are maps of coalgebras

$$\mathbb{k} \xrightarrow{\eta} a \xrightarrow{\varepsilon} \mathbb{k}, \quad \varepsilon \circ \eta = \mathrm{id} \tag{4.16}$$

where ε is the counit, and η is the coaugmentation.

They induce maps of dg Lie algebras:

$$L_0 \xrightarrow{\varepsilon_*} L_a \xrightarrow{\eta_*} L_0 \tag{4.17}$$

such that $\eta_* \circ \varepsilon_* = \mathrm{id}$.

By (4.14), we are interested in

$$F_{X,Y}^f(a) = \{ \theta \in \mathrm{MC}(L_a), \ \eta_*(\theta) = \mathrm{Bar}(f)_{\mathrm{pr}} \}$$

$$(4.18)$$

where $Bar(f)_{pr}$ is considered as a solution of the Maurer-Cartan equation in L_0 .

We identify this MC solution in L_0 with $\theta_0 = \varepsilon_*(\text{Bar}(f)_{\text{pr}})$, regarded as a Maurer-Cartan element in L_a .

Then any Maurer-Cartan element in (4.18) is of form

$$\theta = \theta_0 + \xi \tag{4.19}$$

where

$$\xi \in \operatorname{Ker}(\eta_*) = \operatorname{Hom}_{\Bbbk}(a/\Bbbk, \operatorname{Hom}_{\Sigma}(\operatorname{Bar}_n(X), Y\{n\})[-1]) = L'_a$$
(4.20)

We want to rewrite the Maurer-Cartan equation on $\theta \in L_a$ as some equation on $\xi \in L'_a$. The Maurer-Cartan equation on θ is

$$d\theta + \frac{1}{2}[\theta, \theta] = 0 \iff d_1\xi + [\operatorname{Bar}(f)_{\operatorname{pr}}, \xi] + \frac{1}{2}[\xi, \xi] = 0$$
(4.21)

where $d_1 = d_0 + d_a + d_{Bar}$.

The conclusion is that the Maurer-Cartan equation on θ is the same that the Maurer-Cartan equation on ξ in the dg Lie algebra

$$L_a'' = \left(\operatorname{Hom}_{\Bbbk}(a/\Bbbk, \operatorname{Hom}_{\Sigma}(\operatorname{Bar}_n(X), Y\{n\})[-1]), \ d = d_0 + d_a + d_{\operatorname{Bar}} + \operatorname{ad}(\operatorname{Bar}(f)_{\operatorname{pr}})\right)$$
(4.22)

Note that, as a dg Lie algebra,

$$L''_{a} = \operatorname{Hom}_{\Bbbk}(a/\Bbbk, \operatorname{Def}(X \xrightarrow{f} Y))$$

$$(4.23)$$

Then the adjunction in Proposition 3.7 gives

$$F_{X,Y}^{f}(a) = \{\xi \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a/\Bbbk, \mathrm{Def}(X \xrightarrow{f} Y)))\}$$

= $\mathrm{Hom}_{\mathrm{C}oalg(\mathsf{Comm})}(a, C_{\mathrm{CE}}(\mathrm{Def}(X \xrightarrow{f} Y), \Bbbk))$ (4.24)

4.4.1. Now turn back to our original definition of the functor $F_{X,Y}^f$ as (4.4). It follows immediately that, for a chain of maps of \mathbf{e}_n -algebras in $\mathcal{V}ect(\mathbb{k})_{\Sigma}$

$$X \xrightarrow{f} Y \xrightarrow{g} Z \tag{4.25}$$

there is a map of sets

$$F_{Y,Z}^g(a') \times F_{X,Y}^f(a) \to F_{X,Z}^{gf}(a \otimes a')$$
(4.26)

what gives rise to a map of bifunctors $Coalg \times Coalg \rightarrow Sets$:

$$F_{Y,Z}^g(-_1) \times F_{X,Y}^f(-_2) \to F_{X,Z}^{gf} \circ \bigotimes(-_1, -_2)$$
 (4.27)

Plugging in the representative objects for -1 and -2 and the identity maps, one gets a map of dg cocommutative coalgebras:

$$T_{f,g}: C_{\rm CE}(\operatorname{Def}(Y \xrightarrow{g} Z), \Bbbk) \bigotimes C_{\rm CE}(\operatorname{Def}(X \xrightarrow{f} Y), \Bbbk) \to C_{\rm CE}(\operatorname{Def}(X \xrightarrow{gf} Z), \Bbbk)$$
(4.28)

These maps of dg cocommutative coalgebras are associative:

$$T_{gf,h} \circ (\mathrm{id} \otimes T_{f,g}) = T_{f,hg} \circ (T_{g,h} \otimes \mathrm{id})$$
(4.29)

4.4.2. Plug X = Y = Z, f = g = id to (4.28).

It follows that $A = C_{CE}(\text{Def}(X \xrightarrow{\text{id}} X), \mathbb{k})$ is a monoid object in the category cocommutative dg coalgebras. That is, it is a cocommutative dg bialgebra.

It follows from the Milnor-Moore theorem [Q, Appendix B] that, as a graded Hopf algebra, $A = \mathcal{U}(\mathfrak{g})$, where \mathfrak{g} is the Lie algebra of primitive elements. In our case,

$$\mathfrak{g} = \operatorname{Def}(X \xrightarrow{\operatorname{id}} X)[1] \tag{4.30}$$

Thus, the product (4.28) gives *some* Lie algebra structure on

$$\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)[1] = \prod_{\ell \ge 1} \operatorname{Hom}_{\Sigma_{\ell}}((\mathbf{e}_n^* \circ X\{n\})(\ell), X\{n\}(\ell))$$
(4.31)

It is a Gerstenhaber-like Lie bracket, which has been obtained from the convolution Lie bracket.

4.5. LEMMA. The Lie bracket on $Def(X \xrightarrow{id} X)$, defined in Section 4.2.1, is equal to 0.

PROOF. We know that $C_{CE}(\text{Def}(X \xrightarrow{\text{id}} X, \mathbb{k}) \text{ is a dg Hopf algebra. It is of the form } \mathcal{U}(\mathfrak{g})$, where $\mathfrak{g} = \text{Def}(X \xrightarrow{\text{id}} X)[1]$ is the space of primitive elements. The Chevalley-Eilenberg chain differential is equal to 0 on \mathfrak{g} . As a conclusion, we get that the Chevalley-Eilenberg differential on any power $S^k(\text{Def}(X \xrightarrow{\text{id}} X)[1])$ is 0. It is equivalent to saying that $\text{Def}(X \xrightarrow{\text{id}} X)$ is an abelian Lie algebra.

4.5.1.

4.6. LEMMA. The Lie bracket on $\text{Def}(X \xrightarrow{\text{id}} X)[1]$ given by the Milnor-Moore theorem is identified with the bracket given by the commutator of coderivativations of the cofree e_n^* -coalgebra $\text{Bar}_n(X\{n\})$.

4.7. The case $\mathcal{P} = [X, X]_{\mathcal{Q}}$.

4.7.1. Recall some useful adjunction. Let \mathcal{Q} be an operad, $X, Y \in \mathcal{V}ect(\Bbbk)$ right \mathcal{Q} -modules, $S \in \mathcal{V}ect(\Bbbk)_{\Sigma}$. One has:

$$\operatorname{Hom}_{\Sigma,\operatorname{Mod}-\mathcal{Q}}(S \circ X, Y) = \operatorname{Hom}_{\Sigma}(S, [X, Y]_{\mathcal{Q}})$$

$$(4.32)$$

See Section 2.7.2 for the definition of $[X, Y]_{\mathcal{Q}} \in \mathcal{V}ect(\Bbbk)_{\Sigma}$.

4.7.2. Let \mathcal{Q} be an operad, \mathcal{O} a Koszul operad, X a symmetric sequence in $\mathcal{V}ect(\Bbbk)$, with a $\mathcal{O}-\mathcal{Q}$ -bimodule structure:

$$m\colon \mathcal{O}\circ X\circ \mathcal{Q}\to X\tag{4.33}$$

Recall that a $\mathcal{O}-\mathcal{Q}$ -bimodule structure on a symmetric sequence X gives rise to an operad map:

$$\mathcal{O} \to [X, X]_{\mathcal{Q}} \tag{4.34}$$

See Section 2.7.2 for detail.

4.7.3. Let X be an $\mathcal{O}-\mathcal{Q}$ -bimodule, as above. Consider the bar-complex

$$\operatorname{Bar}_{\mathcal{O}}(X) = (\mathcal{O}^{\mathsf{i}} \circ X, d_{\operatorname{Bar}})$$

The underlying symmetric sequence $\mathcal{O}^{i} \circ X$ is clearly a right \mathcal{Q} -module. It is shown in Section 3.8 that $\operatorname{Bar}_{\mathcal{O}}(X)$ becomes an object of the category $\mathcal{C}(\mathcal{O}^{i}, \mathcal{Q})$, defined there.

The shifted symmetric sequence $X\{n\}$ becomes an $\mathcal{O}\{n\}-\mathcal{Q}\{n\}$ -bimodule.

Consider the case $\mathcal{O} = \mathbf{e}_n$.

Then the complex

$$\operatorname{Bar}_n(X) = (\mathbf{e}_n^* \circ (X\{n\}), d_{\operatorname{Bar}})$$

is an object of the category $\mathcal{C}(\mathsf{e}_n^*, \mathcal{Q}\{n\})$.

Let $f: X \to Y$ be a map of $e_n - Q$ -bimodules, X, Y symmetric sequences. We define a functor

$$_{\mathcal{Q}}F^{f}_{X,Y}$$
: $\operatorname{Coalg} \to \operatorname{Sets}$ (4.35)

as follows:

$${}_{\mathcal{Q}}F^f_{X,Y}(a) = \left\{ \phi \in \operatorname{Hom}_{\mathcal{C}(\mathbf{e}_n^*, \mathcal{Q}\{n\})}(a \star \operatorname{Bar}_n(X), \operatorname{Bar}_n(Y)), \ \phi \circ \eta = \operatorname{Bar}(f) \right\}$$
(4.36)

We use Proposition 3.10. It gives:

$${}_{\mathcal{Q}}F^{f}_{X,Y}(a) = \left\{ \theta \in \mathrm{MC}(\mathrm{Hom}_{\Sigma,\mathrm{Mod}-\mathcal{Q}\{n\}}(a \star \mathrm{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1]), \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}} \right\}$$
(4.37)

One further has:

$$\mathcal{Q}F_{X,Y}^{f}(a) = \left\{ \theta \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a, \mathrm{Hom}_{\Sigma, \mathrm{Mod} - \mathcal{Q}\{n\}}(\mathrm{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1]), \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}} \right\} \\
= \left\{ \theta \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a, \mathrm{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}))), \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}}^{\sim} \right\}$$

$$(4.38)$$

where by $\operatorname{Bar}(f)_{\operatorname{pr}}^{\sim}$ is denoted the element corresponded to $\operatorname{Bar}(f)_{\operatorname{pr}}$ by the adjunction (4.32).

4.7.4. Here we endow $\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})[-1]$ with a dg Lie algebra structure. Consider the symmetric sequence

$$D = \text{Hom}_{lev}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$$
(4.39)

The graded vector space $\operatorname{Hom}_{\Sigma}(\mathbf{e}_n^*, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$ is equal to D_{Σ} .

On the other hand, the operad $[Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}$ acts on $[X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}$ from the left:

$$[Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}} \circ [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}} \to [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}$$
(4.40)

see Section 2.7.2.

Therefore, the convolution operad

$$\operatorname{Hom}_{\mathcal{O}p}(\mathbf{e}_n^*, [Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$$

$$(4.41)$$

acts on D, by Lemma 2.14. Therefore, the same operad acts on the graded space $D_{\Sigma} = \text{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}).$

There is a map of operads

$$\mathbf{e}_n\{n\} \to [Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}$$

Therefore, D and D_{Σ} become algebras over the operad $\operatorname{Hom}_{\mathcal{O}p}(\mathbf{e}_n^*, \mathbf{e}_n\{n\})$.

Now we proceed as in Section 4.2.1. There is a map of operads $\text{Lie}\{1\} \to \text{Hom}_{\mathcal{O}p}(\mathbf{e}_n^*, \mathbf{e}_n\{n\})$, by Lemma 2.16. Therefore, the graded space

$$\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})[-1]$$
(4.42)

is a graded Lie algebra. One easily checks that the differentials d_0 and d_{Bar} differentiate this Lie algebra structure.

Denote

$$Def_0(X,Y)_{\mathcal{Q}} := (Hom_{\Sigma}(\mathbf{e}_n^*, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})[-1], d = d_0 + d_{Bar}^{\sim})$$
(4.43)

We claim that the map $f: X \to Y$ of $\mathbf{e}_n - \mathcal{Q}$ -bimodules defines a degree 1 element in $\mathrm{Def}_0(X,Y)_{\mathcal{Q}}$, satisfying the Maurer-Cartan equation.

Indeed, f defines a map of shifted bar-complexes

$$\operatorname{Bar}(f)\{n\}\colon \operatorname{Bar}_n(X)\to \operatorname{Bar}_n(Y)$$

which is a map of \mathbf{e}_n^* -coalgebras and of right $\mathcal{Q}\{n\}$ -modules. The map $\operatorname{Bar}(f)\{n\}_{\operatorname{pr}}$ is defined by the projection to the cogenerators $\operatorname{Bar}(f)\{n\}_{\operatorname{pr}}$: $\operatorname{Bar}_n(X) \to Y\{n\}$. It is a map of right $\mathcal{Q}\{n\}$ -modules. Now the adjunction (4.32) gives the corresponding degree 0 map in $\operatorname{Hom}_{\Sigma}(\mathbf{e}_n^*, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$. After the shift [-1] it gives a degree 1 element in $\operatorname{Def}_0(X, Y)_{\mathcal{Q}}$. One checks that it satisfies the Maurer-Cartan equation.

Denote the obtained element by $Bar(f)\{n\}_{pr}^{\sim}$, and

$$d_f = \operatorname{ad}(\operatorname{Bar}(f)\{n\}_{\operatorname{pr}}^{\sim})$$

One gets that

$$\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}} := \left(\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})[-1], d = d_{0} + d_{\operatorname{Bar}} + d_{f}\right)$$
(4.44)

is a dg Lie algebra. We call it the Q-relative deformation dg Lie algebra.

Note that

$$[X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}} = ([X, Y]_{\mathcal{Q}})\{n\}$$
(4.45)

Therefore, one has:

$$\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}} = \left(\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{\mathsf{i}}, [X, Y]_{\mathcal{Q}})[-1], d = d_{0} + d_{\operatorname{Bar}} + d_{f}\right)$$
(4.46)

4.7.5. We have:

4.8. PROPOSITION. The functor ${}_{\mathcal{Q}}F^f_{X,Y}$: $\operatorname{Coalg}(\operatorname{Comm}) \to \operatorname{Sets}$ is representable, by the dg coalgebra $a_{\operatorname{rep}} = C_{\operatorname{CE}}(\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}, \Bbbk)$:

$${}_{\mathcal{Q}}F^{f}_{X,Y}(a) = \operatorname{Hom}_{\Sigma,\operatorname{\mathcal{C}oalg}(\mathsf{Comm})}(a, C_{\operatorname{CE}}(\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}, \Bbbk))$$
(4.47)

where $\operatorname{Coalg}(\operatorname{Comm})$ denotes the category of pro-conilpotent cocommutative coalgebras in $\operatorname{Vect}(\Bbbk)_{\Sigma}$, and $C_{\operatorname{CE}}(-)$ stands for the chain Chevalley-Eilenberg complex of a (dg) Lie algebra.

PROOF. The proof is analogous to the proof of Proposition 3.10, and we omit the detail. The only difference is that we use Proposition 3.10 here, instead of Proposition 3.7.

4.8.1. Turn back to the definition of the functor $_{\mathcal{Q}}F_{X,Y}^{f}$, given in (4.36). It follows from this definition that, for a chain of maps of $\mathbf{e}_{n}-\mathcal{Q}$ -bimodules in $\mathcal{V}ect(\mathbb{k})_{\Sigma}$

$$X \xrightarrow{f} Y \xrightarrow{g} Z$$

one gets a map of sets

$${}_{\mathcal{Q}}F^{g}_{Y,Z}(a') \times_{\mathcal{Q}} F^{f}_{X,Y}(a) \to_{\mathcal{Q}} F^{gf}_{X,Z}(a \otimes a')$$

$$(4.48)$$

which gives rise to a map of bifunctors $Coalg \times Coalg \rightarrow Sets$:

$${}_{\mathcal{Q}}F^g_{Y,Z}(-_2) \times_{\mathcal{Q}} F^f_{X,Y}(-_1) \to_{\mathcal{Q}} F^{gf}_{X,Z} \circ \bigotimes(-_1,-_2)$$

$$(4.49)$$

Denote by a_{rep}^f the representing coalgebra for the functor ${}_{\mathcal{Q}}F_{X,Y}^f(-)$. Then (4.48) gives a map:

$$a_{\rm rep}^f \otimes a_{\rm rep}^g \to a_{\rm rep}^{gf}$$
 (4.50)

of dg coalgebras, which enjoys the natural associativity for a chain of four maps $X \to Y \to Z \to W$.

We know that

$$a_{\mathrm{rep}}^f = C_{\mathrm{CE}}(\mathrm{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}, \mathbb{k})$$

Consider the case X = Y, $f = id_X$. Then (4.50) gives a dg bialgebra structure on

$$a_{\operatorname{rep}}^{\operatorname{id}}(X) = C_{\operatorname{CE}}(\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}, \Bbbk)$$

4.9. PROPOSITION. Let X be a $e_n - Q$ -bimodule. Then

$$a_{\operatorname{rep}}^{\operatorname{id}}(X) = C_{\operatorname{CE}}(\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}, \mathbb{k})$$

is a cocommutative dg bialgebra.

One can apply the Milnor-Moore theorem to this cocommutative bialgebra, which gives that, as a graded bialgebra,

$$C_{\rm CE}({\rm Def}(X \xrightarrow{\rm id} X)_{\mathcal{Q}}, \Bbbk) = \mathcal{U}(\mathfrak{g})$$
 (4.51)

where

$$\mathfrak{g} = \operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1]$$

is the space of primitive elements. It becomes a Lie algebra, with the bracket $[x, y] = x * y \mp y * x$.

One has

4.10. LEMMA. The Lie bracket on $\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}$ defined in Section 4.7.4 is equal to 0.

PROOF. The proof is similar to Lemma 4.5, and we omit the detail.

4.11. The case of general \mathcal{P} .

4.11.1. Recall the general definition of the deformation complex $\operatorname{Def}_{\mathfrak{O}p}(\mathcal{O} \xrightarrow{g} \mathcal{P})$ of a morphism of operads. To define $\operatorname{Def}_{\mathfrak{O}p}(\mathcal{O} \xrightarrow{g} \mathcal{P})$, one replaces \mathcal{O} by a cofibrant resolution $R(\mathcal{O})$, and one considers the operad maps $\operatorname{Map}_{\mathfrak{O}p}(R(\mathcal{O}), \mathcal{P})$ in a formal neighborhood of the composition $R(\mathcal{O}) \to \mathcal{O} \xrightarrow{g} \mathcal{P}$. In our case, when \mathcal{O} is Koszul, one chooses $R(\mathcal{O}) = \operatorname{Cobar}_{\mathfrak{O}p}(\mathcal{O}^{i})$.

We get:

$$\operatorname{Map}_{\mathcal{O}p}(R(\mathcal{O}), \mathcal{P}) = \operatorname{MC}\left(\operatorname{Hom}_{\Sigma}(\mathcal{O}^{\mathsf{i}}, \mathcal{P}), d = d_0 + D_g\right)$$
(4.52)

Here we consider the convolution Lie algebra $\operatorname{Hom}_{\Sigma}(\mathcal{O}^{i}, \mathcal{P})$, see Section 2.12.1. The differential component d_{0} comes from the inner differential on \mathcal{P} . The condition of compatibility with the cobar-differential is translated to the Maurer-Cartan equation, see [LV, Section 6.5]. The composition $\operatorname{Cobar}_{\mathcal{O}p}(\mathcal{O}^{i}) \to \mathcal{O} \xrightarrow{g} \mathcal{P}$ defines a Maurer-Cartan element ω_{g} in the convolution Lie algebra, and the differential component $D_{g} = \operatorname{ad}(\omega_{g})$ is defined as the adjoint action of this element.

Finally we define the deformation complex of the operad morphism g as

$$\operatorname{Def}_{\mathfrak{O}p}(\mathcal{O} \xrightarrow{g} \mathcal{P}) = \left(\operatorname{Hom}_{\Sigma}(\mathcal{O}^{i}, \mathcal{P}), d_{0} + D_{g}\right)$$

One has the following general statement:

4.12. LEMMA. Let \mathcal{O} be a Koszul operad, \mathcal{Q} an operad, X an $\mathcal{O}-\mathcal{Q}$ -bimodule. Then the deformation complex $\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1]$ is isomorphic to the deformation complex $\operatorname{Def}(\mathcal{O} \xrightarrow{f} [X, X]_{\mathcal{Q}})$:

$$\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1] \simeq \operatorname{Def}(\mathcal{O} \xrightarrow{f} [X, X]_{\mathcal{Q}})$$
 (4.53)

where the operad map $f: \mathcal{O} \to [X, X]_{\mathcal{Q}}$ is obtained from the bimodule structure on X, as in Section 4.7.2.

PROOF. The isomorphism (4.53) holds for the underlying graded vector spaces. Let us compare the differentials on both sides. The differential components d_0 are equal on the both sides. We claim that, up to the shift by [1],

$$D_g = d_{\text{Bar}} + d_{\text{id}} \tag{4.54}$$

where we use the notations from (4.44). Indeed, D_g looks like the Hochschild cochain differential, it contains the regular and the two extreme terms, corresponded to $\omega_g \circ -$ and $- \circ \omega_g$, correspondingly. These two summands are equal to d_{Bar} and d_{id} , correspondingly.

One can alternatively describe the Lie bracket, given on $\text{Def}(X \xrightarrow{\text{id}} X)_{\mathcal{Q}}[1]$ by the Milnor-Moore theorem [Q, Appendix B], as follows.

4.13. LEMMA. The Lie bracket on $\operatorname{Def}(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1]$ defined via by the Milnor-Moore theorem is equal to the Lie bracket obtained from the convolution pre-Lie bracket on the operadic deformation complex $\operatorname{Def}_{\mathcal{O}p}(\mathbf{e}_n \xrightarrow{g} [X, X]_{\mathcal{Q}})$. For the case $\mathcal{Q} = I$, this Lie bracket can be also described as the Lie bracket on the graded space of coderivations of the cofree \mathbf{e}_n^i -coalgebra $\mathbf{e}_n^i \circ X$.

It is a direct check.

4.13.1. One particular case of previous construction is obtained when $X = \mathcal{Q} = \mathcal{P}$, with the tautological right \mathcal{P} -action on $X = \mathcal{P}$. We know that $[\mathcal{P}, \mathcal{P}]_{\mathcal{P}} = \mathcal{P}$, see Section 2.7.2. We assume we are given a map of operads $g: \mathbf{e}_n \to \mathcal{P}$, which gives, after the operadic shift, a map of operads $g\{n\}: \mathbf{e}_n\{n\} \to \mathcal{P}\{n\}$. Then $\mathcal{P}\{n\}$ becomes a $\mathbf{e}_n^- \mathcal{P}\{n\}$ -bimodule. Lemma 2.8 gives a map of operads

$$\mathbf{e}_n\{n\} \to [\mathcal{P}\{n\}, \mathcal{P}\{n\}]_{\mathcal{P}\{n\}} = \mathcal{P}\{n\}$$

which is equal to the map $g\{n\}$.

Moreover, one easily shows that the deformation complexes

$$\operatorname{Def}_{\mathcal{O}p}(\mathcal{O} \xrightarrow{g} \mathcal{P}) \text{ and } \operatorname{Def}_{\mathcal{O}p}(\mathcal{O}\{n\} \xrightarrow{g\{n\}} \mathcal{P}\{n\})$$

are isomorphic.

Therefore, the results of Section 4.7 are applied to a general map of operads $g: \mathbf{e}_n \to \mathcal{P}$.

- 5. Deformation theory of a morphism of operads $f: \mathbf{e}_n \to \mathcal{P}$ with \mathbf{e}_n coalgebra base
- 5.1. The HOPF ALGEBRA STRUCTURE ON THE OPERAD e_n .

5.1.1. Let \mathcal{E} be a topological operad. Then the diagonal maps $\Delta(n) : \mathcal{E}(n) \to \mathcal{E}(n) \times \mathcal{E}(n)$, which gives rise to a map of operads $\Delta : \mathcal{E} \to \mathcal{E} \times \mathcal{E}$. That is, any topological operad is Hopf.

When we apply the monoidal functor of singular homology, we get an operad $H_{\bullet}(\mathcal{E}, \mathbb{k})$ in graded vector spaces, such that for any n one has a map

$$\Delta(n) \colon H_{\bullet}(\mathcal{E}(n), \Bbbk) \to H_{\bullet}(\mathcal{E}(n), \Bbbk) \otimes H_{\bullet}(\mathcal{E}(n), \Bbbk)$$

It is compatible with the operad compositions, and gives rise to a map of operads

$$\Delta \colon H_{\bullet}(\mathcal{E}, \Bbbk) \to H_{\bullet}(\mathcal{E}, \Bbbk) \otimes H_{\bullet}(\mathcal{E}, \Bbbk)$$

The conclusion is that the homology operad of a topological operad is always a Hopf operad (in graded vector spaces).

5.1.2. It is applied to the operad $\mathbf{e}_n = H_{\bullet}(\mathsf{E}_n, \mathbb{k})$, what makes it a Hopf operad.

One can write down the Hopf structure on \mathbf{e}_n explicitly. The suboperad $\mathsf{Comm} \subset \mathbf{e}_n$ is "group-like": the element $c_s \in \mathsf{Comm}(s) \subset \mathbf{e}_n(s)$, equal to the composition of the product operation $c_2 \in \mathsf{Comm}(2)$, satisfies

$$\Delta(c_s) = c_s \otimes c_s \tag{5.1}$$

The suboperad $\text{Lie}\{-n+1\} \subset \mathbf{e}_n$ forms the subspace of primitive elements in each arity, in the sense that

$$\Delta(\ell_s) = \ell_s \otimes c_s + c_s \otimes \ell_s \tag{5.2}$$

for any $\ell_s \in \text{Lie}\{-n+1\}(s)$.

As $\mathbf{e}_n = \mathsf{Comm} \circ \mathsf{Lie}\{-n+1\}$, the condition that Δ is an operad map defines it on any component $\mathbf{e}_n(s)$.

5.2. LEMMA. Consider the map of operads

$$\mathsf{Lie}\{1\} \to \mathsf{e}_n\{n\} \xrightarrow{\mathrm{Hopf}} \mathsf{e}_n\{n\} \otimes_{lev} \mathsf{e}_n = \mathrm{Hom}_{lev}(\mathsf{e}_n^*\{-n\}, \mathsf{e}_n) = \mathrm{Hom}_{lev}(\mathsf{e}_n^i, \mathsf{e}_n)$$
(5.3)

where the leftmost map is the shifted map $\text{Lie}\{-n+1\} \rightarrow e_n$. Then the composition is equal to the map defined in Lemma 2.16.

PROOF. We deal with two a priori different maps $\text{Lie}\{1\} \rightarrow \mathcal{O}$ to an operad \mathcal{O} . The operad Lie is quadratic, therefore the two maps coincide if they coincide on $\text{Lie}\{1\}(2)$. But for arity 2 it is checked by an explicit computation, using (5.1) and (5.2), on one side, and (2.56), on the other side.

5.2.1. An immediate consequence of the Hopf structure on the operad e_n is:

5.3. LEMMA. Let b_1, b_2 be two \mathbf{e}_n -algebras. Then the tensor product of the underlying (dg) vector spaces $b_1 \otimes b_2$ is naturally a \mathbf{e}_n -algebra. Similarly, let a_1, a_2 be two \mathbf{e}_n^* -coalgebras. Then the tensor product of the underlying (dg) vector spaces is naturally a \mathbf{e}_n^* -coalgebra.

PROOF. We prove the first statement, the second is analogous. For x an algebra over an operad \mathcal{O}_1 and y an algebra over an operad \mathcal{O}_2 , the tensor product $x \otimes y$ is an algebra over the operad $\mathcal{O}_1 \otimes \mathcal{O}_2$. In our case, $b_1 \otimes b_2$ is an algebra over the operad $\mathbf{e}_n \otimes \mathbf{e}_n$. The diagonal map $\Delta : \mathbf{e}_n \to \mathbf{e}_n \otimes \mathbf{e}_n$ makes $b_1 \otimes b_2$ an algebra over \mathbf{e}_n .

We will use the following variation of the lemma:

5.4. LEMMA. Let $b \in \operatorname{Vect}(\mathbb{k}), X \in \operatorname{Vect}(\mathbb{k})_{\Sigma}$ be algebras over the operad \mathbf{e}_n . Then $b \star X = b^{(0)} \boxtimes X$ is an algebra over the operad \mathbf{e}_n . Similarly, for an \mathbf{e}_n^* -coalgebra $a \in \operatorname{Vect}(\mathbb{k})$, and an \mathbf{e}_n^* -coalgebra $Y \in \operatorname{Vect}(\mathbb{k})_{\Sigma}$, the symmetric sequence $a \star Y$ is a coalgebra over \mathbf{e}_n^* .

See Section 4.1.1 for the definition of $a \star X$. The statement is proven analogously with the Lemma above, using Lemma 4.2.

5.5. The functor $_{\mathcal{Q}}G^f_{X,Y}$ and its representability.

5.5.1. Let X be an e_n -algebra in $\mathcal{V}ect(\mathbb{k})_{\Sigma}$. Then the bar-complex

$$\operatorname{Bar}_n(X) = (\mathbf{e}_n^*(X\{n\}), \ d_{\operatorname{Bar}})$$

is an e_n^* -coalgebra, see (4.2), (4.3).

Let a be another e_n^* -coalgebra. Then $a \star Bar_n(X)$ is a e_n -coalgebra, by Lemma 5.4. We consider directly the "relative" case here.

Let \mathcal{Q} be an operad, X, Y-two $\mathbf{e}_n - \mathcal{Q}$ -bimodules, $f: X \to Y$ a bimodule map. The map f defines a map $\operatorname{Bar}(f): \operatorname{Bar}_n(X) \to \operatorname{Bar}_n(Y)$. It is a morphism in the category $\mathcal{C}(\mathbf{e}_n^*, \mathcal{Q}\{n\})$, see Section 3.8.

Denote by $\operatorname{Coalg}_n = \operatorname{Coalg}_n(\Bbbk)$ the category of pro-conilpotentent \mathbf{e}_n^* -coalgebras over \Bbbk .

Define the functor

$$_{\mathcal{Q}}G^f_{X,Y}$$
: $\mathfrak{Coalg}_n \to \mathfrak{Sets}$

as follows:

$${}_{\mathcal{Q}}G^{f}_{X,Y}(a) = \left\{ \phi \in \operatorname{Hom}_{\mathfrak{C}(\mathbf{e}_{n}^{*},\mathcal{Q}\{n\})}(a \star \operatorname{Bar}_{n}(X), \operatorname{Bar}_{n}(Y)), \ \phi \circ \eta = \operatorname{Bar}(f) \right\}$$
(5.4)

Here $\eta \colon \mathbb{k} \to a$ is the coaugmentation map.

Proposition 3.10 gives:

$$\mathcal{Q}G_{X,Y}^{f}(a) = \{\theta \in \mathrm{MC}(\mathrm{Hom}_{\Sigma,\mathrm{Mod}-\mathcal{Q}\{n\}}((a \star \mathrm{Bar}_{n}(X))/\Bbbk, Y\{n\})[-1], \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}}\} \\ = \{\theta \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a, \mathrm{Hom}_{\Sigma,\mathrm{Mod}-\mathcal{Q}\{n\}}(\mathrm{Bar}_{n}(X)/\Bbbk, Y\{n\})[-1]), \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}}\} \\ = \{\theta \in \mathrm{MC}(\mathrm{Hom}_{\Bbbk}(a, \mathrm{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}))), \ \theta \circ \eta = \mathrm{Bar}(f)_{\mathrm{pr}}^{\sim}\}$$

$$(5.5)$$

5.5.2. Denote

$$\operatorname{Def}_{0}^{n}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1] = \operatorname{Def}_{0}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1] = (\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}), d = d_{0} + d_{\operatorname{Bar}})$$
(5.6)
cf. (4.43).

5.6. LEMMA. In the notations as above, the complex $\operatorname{Def}_0^n(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]$ enjoys a natural $e_n\{n\}$ -algebra structure. Moreover, the underlying Lie $\{1\}$ -algebra structure on $\operatorname{Def}_0^n(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]$ agrees with the Lie algebra structure on $\operatorname{Def}_0(X \xrightarrow{f} Y)_{\mathcal{Q}}$ given in Section 4.7.4.

PROOF. By Lemma 2.14, the convolution operad $\operatorname{Hom}_{lev}(\mathbf{e}_n^*, [Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$ acts on the symmetric sequence $\operatorname{Hom}_{lev}(\mathbf{e}_n^*, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$ and, therefore, on

$$\operatorname{Hom}_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$$

This action is compatible with the differential $d_0 + d_{Bar}^{\sim}$, which, in turn, acts on the components of the level Hom. Now we recall the operad map $\mathbf{e}_n \to [Y, Y]_{\mathcal{Q}}$, coming from the $\mathbf{e}_n - \mathcal{Q}$ -bimodule structure, see Section 4.7.2. Then there is the shifted operad map $\mathbf{e}_n^*\{n\} \to [Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}}$, and the convolution operad $\operatorname{Hom}_{lev}(\mathbf{e}_n^*, \mathbf{e}_n\{n\}) = \mathbf{e}_n \otimes \mathbf{e}_n\{n\}$ becomes acting on $\operatorname{Def}_0^n(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]$. Now the Hopf structure on the operad \mathbf{e}_n (see Section 5.1.2) gives an operad map $\mathbf{e}_n\{n\} \to \mathbf{e}_n \otimes \mathbf{e}_n\{n\}$, which proves the first statement.

For the second statement, we notice that the Lie algebra structure on $\operatorname{Def}_0(X \xrightarrow{f} Y)_{\mathcal{Q}}$ given in Section 4.7.4 was constructed in the similar way, considering the action of the operad $\operatorname{Hom}_{lev}(\mathbf{e}_n^*, [Y\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})$ as the first step, followed by restricting this action to the action of operad $\operatorname{Hom}_{lev}(\mathbf{e}_n^*, \mathbf{e}_n\{n\})$, and then restricting it to the operad $\operatorname{Lie}\{1\}$, by Lemma 2.16. So the claim is that the two $\operatorname{Lie}\{1\}$ actions coincide, and it follows from Lemma 5.2.

5.6.1. We can twist the differential of deformation complex $\operatorname{Def}_0^n(X \xrightarrow{f} Y)_{\mathcal{Q}}$ (which, as a dg Lie algebra, does not depend on the map f) by the adjoint action $d_f = \operatorname{ad}(\operatorname{Bar}(f)\{n\}_{\operatorname{pr}}^{\sim})$ of the Maurer-Cartan element $\operatorname{Bar}(f)\{n\}_{\operatorname{pr}}^{\sim}$, as in Section 4.7.4:

$$Def^{n}(X \xrightarrow{f} Y)_{\mathcal{Q}} := \left(Hom_{\Sigma}(\mathbf{e}_{n}^{*}, [X\{n\}, Y\{n\}]_{\mathcal{Q}\{n\}})[-1], d = d_{0} + d_{Bar} + d_{f}\right)$$
(5.7)

The new thing is that this twisting preserves the operad $\mathbf{e}_n\{n\}$ acting on the shifted by [1] complex $\operatorname{Def}^n(X \xrightarrow{f} Y)_{\mathcal{O}}[1]$.

It can be checked by hand, of course. A more conceptual explanation goes through the second statement of Lemma 5.6. By this statement, the operad acting on the twisted complex is the operadic twisting $\text{Tw}(\text{Lie}\{1\} \rightarrow e_n\{n\})$, see [DW]. This operad is wellknown to be weak equivalent to the operad $e_n\{n\}$, see loc.cit., Sect. 4.3.

We conclude, that $\operatorname{Def}^n(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]$, defined in (5.7), is a dg $\mathbf{e}_n\{n\}$ -algebra.

5.7. REMARK. Note that $\operatorname{Def}_0^n(X \xrightarrow{f} Y)_{\mathcal{Q}}$ is well-defined as a symmetric sequence, and only the last differential component d_f mixes the components of the symmetric sequence up. As well, the operadic twisting is a useful tool to be applied at this "last step", when the components of the symmetric sequence from the previous steps are being mixed up, and we are interesting which operad acts on the resulting complex. Many non-trivial examples of this idea in use can be found in [W], [CW], [DW], ...

5.7.1. Consider the bar-complex $\operatorname{Bar}_{e_n\{n\}}(\operatorname{Def}^n(X \xrightarrow{f} Y)_{\mathcal{Q}}[1])$ over the operad $e_n\{n\}$. It is a e_n^* -coalgebra.

One has:

5.8. PROPOSITION. The functor $_{\mathcal{Q}}G_{X,Y}^{f}$: $\operatorname{Coalg}_{n} \to \operatorname{Sets}$ from the category of pro-conilpotent \mathbf{e}_{n}^{*} -coalgebras to the category of sets is representable, by the \mathbf{e}_{n}^{*} -coalgebra $A_{\operatorname{rep}}^{f} = \operatorname{Bar}_{\mathbf{e}_{n}\{n\}}(\operatorname{Def}^{n}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]):$

$$\mathcal{Q}G_{X,Y}^{f}(a) = \operatorname{Hom}_{\mathcal{C}(\mathbf{e}_{n}^{*},\mathcal{Q})}(a \star \operatorname{Bar}_{n}(X), \operatorname{Bar}_{n}(Y))$$
$$= \operatorname{Hom}_{\mathcal{C}oalg_{n}}(a, \operatorname{Bar}_{\mathbf{e}_{n}\{n\}}(\operatorname{Def}^{n}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1]))$$
(5.8)

PROOF. We start with the last line of (5.5), and then continue as in the proof of Proposition 4.4, using the adjunction given in Proposition 4.3. We skip the detail, as the argument is analogous to the proof of Proposition 4.4 (which is, by its own, borrowed from [T2, Prop. 3.2]).

5.9. Turn back to the definition of the functor $_{\mathcal{Q}}G_{X,Y}^{f}$, given in (5.4). It follows from this definition that, for a chain of maps of $\mathbf{e}_{n}-\mathcal{Q}$ -bimodules in $\mathcal{V}ect(\mathbf{k})_{\Sigma}$

$$X \xrightarrow{f} Y \xrightarrow{g} Z$$

one gets a map of sets

$${}_{\mathcal{Q}}G^g_{Y,Z}(a') \times_{\mathcal{Q}} G^f_{X,Y}(a) \to_{\mathcal{Q}} G^{gf}_{X,Z}(a \otimes a')$$
(5.9)

which gives rise to a map of bifunctors $\mathbb{C}oalg_n\times\mathbb{C}oalg_n\to\mathbb{S}ets$:

$${}_{\mathcal{Q}}G^g_{Y,Z}(-_2) \times_{\mathcal{Q}} G^f_{X,Y}(-_1) \to_{\mathcal{Q}} G^{gf}_{X,Z} \circ \bigotimes(-_1,-_2)$$
(5.10)

Denote by A_{rep}^f the representing *n*-coalgebra for the functor ${}_{\mathcal{Q}}G_{X,Y}^f(-)$. Then (4.48) gives a map:

$$A_{\rm rep}^f \otimes A_{\rm rep}^g \to A_{\rm rep}^{gf} \tag{5.11}$$

of dg coalgebras, which enjoys the natural associativity for a chain of four maps $X \to Y \to Z \to W$.

We know from Proposition 5.8 that

$$A_{\mathrm{rep}}^{f} = \mathrm{Bar}_{\mathbf{e}_{n}\{n\}}(\mathrm{Def}^{n}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1])$$

Consider the case X = Y, $f = id_X$. Then (5.11) gives a monoid in the category of dg *n*-coalgebras structure on $A^{id}_{rep}(X) = \operatorname{Bar}_{e_n\{n\}}(\operatorname{Def}^n(X \xrightarrow{id} X)_{\mathcal{Q}})$:

5.10. PROPOSITION. Let X be a $e_n - Q$ -bimodule. Then

$$A_{\operatorname{rep}}^{\operatorname{id}}(X) = \operatorname{Bar}_{\mathbf{e}_n\{n\}}(\operatorname{Def}^n(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1])$$

is a monoid in the category of e_n^* -coalgebras.

5.11. DEFINITION. Let A be a \mathbf{e}_n^* -coalgebra, endowed with a monoid structure, that is, with a map

$$A \otimes A \to A \tag{5.12}$$

of \mathbf{e}_n^* -coalgebras, which is associative, and has a unit given by the coaugmentation map of A. Then we say that A is $a \mathbf{e}_n^*$ -bialgebra.

Proposition 5.8 says that the e_n^* -coalgebra

$$A_{\operatorname{rep}}^{\operatorname{id}}(X) = \operatorname{Bar}_{\mathbf{e}_n\{n\}}(\operatorname{Def}^n(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1])$$

is a e_n^* -bialgebra.

We know from Lemma 4.12 that, as a complex, $\operatorname{Def}^n(X \xrightarrow{\operatorname{id}} X)_{\mathcal{Q}}[1] = \operatorname{Def}_{\mathfrak{O}p}(\mathbf{e}_n \to [X, X]_{\mathcal{Q}}).$

Then Proposition 5.10 gives:

5.12. COROLLARY. Let X be a $\mathbf{e}_n\{n\} - \mathcal{Q}$ -bimodule, $g: \mathbf{e}_n\{n\} \to [X, X]_{\mathcal{Q}}$ the associated map. Then the operadic deformation complex $\mathrm{Def}_{\mathbb{O}p}(\mathbf{e}_n\{n\} \xrightarrow{g} [X, X]_{\mathcal{Q}})$ is a $\mathbf{e}_n\{n\}$ -algebra, and $\mathrm{Bar}_{\mathbf{e}_n\{n\}}(\mathrm{Def}_{\mathbb{O}p}(\mathbf{e}_n\{n\} \xrightarrow{g} [X, X]_{\mathcal{Q}}))$ has a \mathbf{e}_n^* -bialgebra structure.

One particular case of previous construction is obtained when $X = \mathcal{Q} = \mathcal{P}$, with the tautological right \mathcal{P} -action on $X = \mathcal{P}$. We know that $[\mathcal{P}, \mathcal{P}]_{\mathcal{P}} = \mathcal{P}$, see Section 2.7.2. We assume we are given a map of operads $f: \mathbf{e}_n \to \mathcal{P}$, then $f\{n\}: \mathbf{e}_n\{n\} \to \mathcal{P}\{n\}$, which makes $\mathcal{P}\{n\}$ a $\mathbf{e}_n\{n\}-\mathcal{P}\{n\}$ -bimodule. Lemma 2.8 gives a map of operads

$$\mathbf{e}_n\{n\} \to [\mathcal{P}\{n\}, \mathcal{P}\{n\}]_{\mathcal{P}\{n\}} = \mathcal{P}\{n\}$$

which is equal to the map $f\{n\}$. We mention that $\operatorname{Def}_{\mathfrak{O}p}(\mathcal{O} \xrightarrow{f} \mathcal{P}) = \operatorname{Def}_{\mathfrak{O}p}(\mathcal{O}\{k\} \xrightarrow{f\{k\}} \mathcal{P}\{k\})$, for any k.

One gets:

5.13. COROLLARY. Let $f: \mathbf{e}_n \to \mathcal{P}$ be an operad map. Then $\operatorname{Def}_{\mathfrak{O}p}(\mathbf{e}_n \xrightarrow{f} \mathcal{P})$ is a $\mathbf{e}_n\{n\}$ -algebra, and $\operatorname{Bar}_{\mathbf{e}_n\{n\}}(\operatorname{Def}_{\mathfrak{O}p}(\mathbf{e}_n \xrightarrow{f} \mathcal{P}))$ is a \mathbf{e}_n^* -bialgebra.

In the next Section, we link the dg cocommutative bialgebra $a_{\text{rep}}^{\text{id}}(X)$ and the dg e_n^* -bialgebra $A_{\text{rep}}^{\text{id}}(X)$, and investigate the higher structure we have found.

6. A proof of Theorem 1.2

Section 6.1 just reproduces arguments from [T2]. For convenience of the reader, we briefly recall them here. It shows that $\text{Def}_{\mathcal{O}p}(\mathbf{e}_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\})[-n]$ is a homotopy (n + 1)-algebra. Note that here we only use the deformation theory with \mathbf{e}_n -coalgebra base, see Section 5.

In Section 6.2, we give an explicit description of the underlying (homotopy) Lie bracket on $\operatorname{Def}_{\mathbb{O}p}(\mathbf{e}_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\})[-n][n] = \operatorname{Def}_{\mathbb{O}p}(\mathbf{e}_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\})$ of this homotopy \mathbf{e}_{n+1} algebra as the strict operadic convolution Lie bracket. It is proven by comparison of the representation objects $a_{\operatorname{rep}}(X)$ and $A_{\operatorname{rep}}(X)$ for the deformation theories with cocommutative and \mathbf{e}_n -coalgebra bases, correspondingly. Our argument here is hopefully somewhat more transparent than the one in [T2].

6.1. Let $f: \mathbf{e}_n \to \mathcal{P}$ be a morphism of operads; it defines the shifted map of operads $f\{n\}: \mathbf{e}_n\{n\} \to \mathcal{P}\{n\}$. The deformation complex $\operatorname{Def}_{\mathcal{O}p}(\mathbf{e}_n \xrightarrow{f} \mathcal{P}) = \operatorname{Def}_{\mathcal{O}p}(\mathbf{e}_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\})$ enjoys a structure of $\mathbf{e}_n\{n\}$ -algebra, by Lemma 5.6 and the discussion thereafter.

Consider the bar-complex $\operatorname{Bar}_{e_n\{n\}}(\operatorname{Def}_{\mathcal{O}p}(e_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\}))$ which is an e_n^* -coalgebra. In general, the underlying complex of $\operatorname{Bar}_{e_n\{n\}}(X)$ is $e_n^* \circ X$. As $e_n = \operatorname{Comm} \circ \operatorname{Lie}\{-n+1\}$, the underlying graded space of

$$\mathbf{e}_n^* \circ X = \mathsf{Comm}^* \circ (\mathsf{Lie}^*\{n-1\} \circ X) \tag{6.1}$$

Denote $X = \operatorname{Def}_{\mathfrak{O}p}(\mathsf{e}_n\{n\} \xrightarrow{f\{n\}} \mathcal{P}\{n\}).$

Corollary 5.13 says that $\operatorname{Bar}_{e_n\{n\}}(X)$ is a e_n^* -bialgebra. Consider the underlying cocommutative bialgebra. Then the Milnor-Moore theorem [Q, Appendix B] gives a Lie algebra structure on the graded space of primitive elements. The space of primitive elements is $\operatorname{Lie}^*\{n-1\} \circ X$. In general, if we have a dg cocommutative coalgebra, the space of primitive elements is a subcomplex. The space of primitive elements is closed under the bracket $ab \mp ba$; therefore, $\operatorname{Lie}^*\{n-1\} \circ X$ becomes a Lie algebra. The Lie algebra structure on $\operatorname{Lie}^*\{n-1\} \circ X$ is compatible with the cofree $\operatorname{Lie}^*\{n-1\}$ -coalgebra structure, by

$$\delta([x,y]) = [\delta x, y] + (-1)^{|x|(d-1)}[x, \delta y]$$
(6.2)

where δ is the Lie^{*}{n-1}-cobracket.

The Chevalley-Eilenberg chain complex of the Lie algebra

$$\mathsf{Lie}^*\{n-1\} \circ X$$

is $\mathsf{Comm}^*\{-1\} \circ \mathsf{Lie}^*\{n-1\} \circ X$ The identity (6.2) is translated to the statement that the Chevalley-Eilenberg differential differentiates the $\mathsf{Lie}^*\{n-1\}$ -coalgebra structure, making $\mathsf{Comm}^*\{-1\} \circ \mathsf{Lie}^*\{n-1\} \circ X$ a dg $\mathsf{Comm}^*\{-1\} \circ \mathsf{Lie}^*\{n-1\}$ -coalgebra.

One has:

$$\mathsf{Comm}^*\{-1\} \circ \mathsf{Lie}^*\{n-1\} \circ X = (\mathsf{Comm}^*\{-n-1\} \circ \mathsf{Lie}^*\{-1\} \circ X\{-n\})\{n\}$$
$$= (\mathsf{e}^*_{n+1}\{-n-1\} \circ X[-n])[n]$$

As $(\operatorname{Comm}^{*}\{-1\} \circ \operatorname{Lie}^{*}\{n-1\} \circ X, d)$ is a dg coalgebra over the cooperad $\operatorname{Comm}^{*}\{-1\} \circ \operatorname{Lie}^{*}\{n-1\}$, the shifted by [-n] complex $\operatorname{Comm}^{*}\{-n-1\} \circ \operatorname{Lie}^{*}\{-1\} \circ X\{-n\}$ is a coalgebra over the cooperad $(\operatorname{Comm}^{*}\{-1\} \circ \operatorname{Lie}^{*}\{n-1\})\{-n\} = e_{n+1}^{*}\{-n-1\}$.

Here have we repeatedly used Lemma 2.9(i).

The conclusion is that $e_{n+1}^* \{-n-1\} \circ (X[-n])$ is a dg coalgebra over $e_{n+1}^* \{-n-1\}$. By definition, it means that X[-n] is a homotopy (n+1)-algebra.

6.2. It remains to prove the second claim of Theorem 1.2, describing the Lie bracket of degree -n in terms of the operadic convolution bracket.

6.2.1. The idea is to link the commutative coalgebra $a_{\text{rep}}(f)$ representing the functor $_{\mathcal{Q}}F_{X,Y}^{f}$ with the *n*-coalgebra $A_{\text{rep}}(f)$ representing the functor $_{\mathcal{Q}}G_{X,Y}^{f}$.

There is the inclusion functor $i: Coalg \to Coalg_n$. It admits a right adjoint $R: Coalg_n \to Coalg$, defined as follows. The cocommutative coalgebra R(a) is defined as the biggest cocommutative coalgebra contained in Ker δ , where δ is the cobracket (see [T2, Prop. 4.4]).

Therefore, for any $a \in Coalg$, one has:

$$\operatorname{Hom}_{\operatorname{Coalg}_n}(i(a), A_{\operatorname{rep}}(f)) = \operatorname{Hom}_{\operatorname{Coalg}}(a, R(A_{\operatorname{rep}}(f)))$$
(6.3)

On the other hand,

$$\operatorname{Hom}_{\operatorname{Coalg}_n}(i(a), A_{\operatorname{rep}}(f)) =_{\mathcal{Q}} G^f_{X,Y}(i(a)) =_{\mathcal{Q}} F^f_{X,Y}(a) = \operatorname{Hom}_{\operatorname{Coalg}}(a, a_{\operatorname{rep}}(f))$$
(6.4)

It follows from the Yoneda lemma that

$$R(A_{\rm rep}(f)) = a_{\rm rep}(f) \tag{6.5}$$

The functor R is lax-monoidal (it follows either from an explicit computation, or from the general fact that a right adjoint to a colax-monoidal functor is lax-monoidal; the functor i is strict monoidal and in particular colax-monoidal). Moreover, (6.5) is compatible with the composition properties

$$a_{\rm rep}(f) \otimes a_{\rm rep}(g) \to a_{\rm rep}(gf)$$

and

$$A_{\rm rep}(f) \otimes A_{\rm rep}(g) \to A_{\rm rep}(gf)$$

for a chain of morphisms $X \xrightarrow{f} Y \xrightarrow{g} Z$, in the sense that the diagram

commutes, where the left vertical arrow is the lax monoidal map. It is straightforward.

Thus we get:

6.3. PROPOSITION. Let X be a $\mathbf{e}_n - \mathcal{Q}$ -bimodule. Denote by $a_{\mathrm{rep}}(\mathrm{id}_X)$ (corresp., $A_{\mathrm{rep}}(\mathrm{id}_X)$) the commutative coalgebra (corresp., the n-coalgebra), representing the functor $\mathcal{Q}F_{X,X}^{\mathrm{id}}$ (corresp., $\mathcal{Q}G_{X,X}^{\mathrm{id}}$). Then the functor R defined above yields a map $R: A_{\mathrm{rep}}(\mathrm{id}_X) \to a_{\mathrm{rep}}(\mathrm{id}_X)$ which is a map of monoid objects.

6.3.1. In the case of a cofree *n*-coalgebra $e_n^* \circ V$, the result of application of the functor R is the cofree cocommutative coalgebra cogenerated by V:

$$R(\mathbf{e}_n^* \circ V) = \mathsf{Comm}^* \circ V \tag{6.7}$$

In our case

$$A_{\rm rep}(f) = \operatorname{Bar}_{\mathbf{e}_n\{n\}}(\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}) = \left(\mathbf{e}_n^* \circ (\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}), d\right)$$
(6.8)

and

$$a_{\rm rep}(f) = \operatorname{Bar}_{\mathsf{Lie}\{1\}}(\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}) = \left(\operatorname{\mathsf{Comm}}^* \circ (\operatorname{Def}(X \xrightarrow{f} Y)_{\mathcal{Q}}[1])), d'\right) \tag{6.9}$$

We have a priori two different Lie algebra structures on $\text{Def}(X \xrightarrow{\text{id}} X)_{\mathcal{Q}}[1]$, defined from the monoid structures on $a_{\text{rep}}(\text{id}_X)$ and on $A_{\text{rep}}(\text{id}_X)$. Proposition 6.3 and the computation thereafter imply that these two Lie algebra structures are equal. On the other hand, we know the one obtained on $a_{\text{rep}}(\text{id}_X)$, from Lemma 4.13. It completes the proof of Theorem 1.2.

References

- [Ba1] M.A. Batanin, The symmetrization of *n*-operads and compactification of real configuration spaces, Advances in Math. **211**(2007), 684-725
- [Ba2] M.A. Batanin, The Eckmann-Hilton argument and higher operads, Advances in Math., 217(2008), 334-385
- [BB] M.A. Batanin, C. Berger, The lattice path operad and Hochschild cochains, *Contemp. Math.* **504**(2009), 23-52
- [BF] C. Berger, B. Fresse, Combinatorial operad actions on cochains, Proc. Math. Cambridge Philos. Soc., 137(1)(2004), 135-174
- [BFSV] C. Baltenu, Z. Fiedorowicz, R. Schwänzl, R. Vogt, Iterated Monoidal Categories, Advances in Math. 176(2003), 277–349
- [BM] M.A. Batanin, M. Markl, Operadic categories and Duoidal Deligne's conjecture, Advances in Math. 285(2015), 1630-1687
- [CW] D. Calaque, T. Willwacher, Triviality of the higher formality theorem, Proceedings of the AMS, vol. 143, no. 12, 2015, 5181–5193
- [DW] V. Dolgushev, T. Willwacher, Operadic twisting–with an application to Deligne's conjecture, J. Pure Appl. Algebra 219 (2015), no. 5, 1349-1428

- [F1] B. Fresse, Modules over operads and functors, Springer LNM 1967
- [F2] B. Fresse, Homotopy of Operads and Grothendieck-Teichmller Groups, AMS Mathematical Surveys and Monographs 217, AMS 2017 (in two volumes)
- [FTW] B. Fresse, V. Turchin, T. Willwacher, The rational homotopy of mapping spaces of E_n operads, preprint arXiv: 1703.06123
- [FW] B. Fresse, T. Willwacher, The intrinsic formality of E_n -operads, to appear in Journal of the EMS
- [GJ] E. Getzler, J.D.S. Jones, Operads, homotopy algebra and iterated integrals for double loop spaces, preprint arXiv:hep-th/9403055
- [GK] V. Ginzburg, M. Kapranov, Koszul duality for operads, Duke Math. J. 76 (1994), no. 1, 203-272
- [GY] G. Ginot, S. Yalin, Deformation theory of bialgebras, higher Hochschild cohomology and formality, arXiv: 1606.01504
- [K1] M. Kontsevich, Formal (non)commutative symplectic geometry, The Gelfand Mathematical Seminars, 1990-1992, 173-187, Birkhuser Boston, Boston, MA, 1993
- [K2] M. Kontsevich, Deformation quantization of Poisson manifolds, Lett. Math. Phys. 66 (2003), no. 3, 157-216
- [K3] M. Kontsevich, Operads and motives in deformation quantization. Mosh Flato (19371998), Lett. Math. Phys. 48 (1999), no. 1, 35-72.
- [K4] M. Kontsevich, Derived Grothendieck-Teichmüller group and graph complexes [after T. Willwacher], Bourbaki talk January 2017
- [KM] M. Kapranov, Yu. Manin, Modules and Morita theory for operads, American Journal of Mathematics 123 (2001), 811-838
- [KS] M. Kontsevich, Y. Soibelman, Deformations of algebras over operads and the Deligne conjecture, *Confrence Mosh Flato 1999*, Vol. I (Dijon), 255-307, Math. Phys. Stud., 21, Kluwer Acad. Publ., Dordrecht, 2000
- [Leg] F. De Leger, Operads and embeddings, preprint Macquarie University, Australia
- [LV] J.-L. Loday, B. Vallette, Algebraic operads, Grundlehren der Mathematischen Wissenschaften, 346, Springer, 2012
- [MS1] J. McClure, J. Smith, A solution of Deligne's conjecture, In Recent progress in homotopy theory (Baltimore, MD, 2000), Contemp. Math., 293(2002), 153-193
- [MS2] J. McClure, J. Smith, Multivariable cochain operations and little n-cubes, J. Amer. Math. Soc., 16(3)(2003), 681-704
- [Q] D. Quillen, Rational homotopy theory, Annals of Mathematics no. 90 (1969), 205295
- [R] Ch. Rezk, Spaces of algebra stuctures and cohomology of operads, MIT Ph.D. Thesis
- [St] A.E. Stanculescu, A remark on the Morita theorem for operads, Arch. Math. (Brno) 47 (2011), no. 2, 139-150

- [T1] D. Tamarkin, Another proof of M. Kontsevich formality theorem, preprint arXiv:math/9803025
- [T2] D. Tamarkin, Deformation complex of a d-algebra is a (d + 1)-algebra, preprint arXiv:math/0010072
- [T3] D. Tamarkin, Action of the Grothendieck-Teichmueller group on the operad of Gerstenhaber algebras, preprint arXiv:math/0202039
- [T4] D. Tamarkin, What do dg categories form? Compos. Math. 143 (2007), no. 5, 1335–1358
- [W] T. Willwacher, M. Kontsevich's graph complex and the Grothendieck-Teichmller Lie algebra, Invent. Math. 200 (2015), no. 3, 671-760

Universiteit Antwerpen, Campus Middelheim, Wiskunde en Informatica, Gebouw G Middelheimlaan 1, 2020 Antwerpen, België

Laboratory of Algebraic Geometry, National Research University Higher School of Economics,

Moscow, Russia Email: Boris.Shoikhet@uantwerpen.be

This article may be accessed at http://www.tac.mta.ca/tac/

THEORY AND APPLICATIONS OF CATEGORIES will disseminate articles that significantly advance the study of categorical algebra or methods, or that make significant new contributions to mathematical science using categorical methods. The scope of the journal includes: all areas of pure category theory, including higher dimensional categories; applications of category theory to algebra, geometry and topology and other areas of mathematics; applications of category theory to computer science, physics and other mathematical sciences; contributions to scientific knowledge that make use of categorical methods.

Articles appearing in the journal have been carefully and critically refereed under the responsibility of members of the Editorial Board. Only papers judged to be both significant and excellent are accepted for publication.

SUBSCRIPTION INFORMATION Individual subscribers receive abstracts of articles by e-mail as they are published. To subscribe, send e-mail to tac@mta.ca including a full name and postal address. Full text of the journal is freely available at http://www.tac.mta.ca/tac/.

INFORMATION FOR AUTHORS LATEX2e is required. Articles may be submitted in PDF by email directly to a Transmitting Editor following the author instructions at http://www.tac.mta.ca/tac/authinfo.html.

MANAGING EDITOR. Robert Rosebrugh, Mount Allison University: rrosebrugh@mta.ca

 $T_EXNICAL$ EDITOR. Michael Barr, McGill University: barr@math.mcgill.ca

ASSISTANT T_EX EDITOR. Gavin Seal, Ecole Polytechnique Fédérale de Lausanne: gavin_seal@fastmail.fm

TRANSMITTING EDITORS.

Clemens Berger, Université de Nice-Sophia Antipolis: cberger@math.unice.fr Julie Bergner, University of Virginia: jeb2md (at) virginia.edu Richard Blute, Université d'Ottawa: rblute@uottawa.ca Gabriella Böhm, Wigner Research Centre for Physics: bohm.gabriella (at) wigner.mta.hu Valeria de Paiva: Nuance Communications Inc: valeria.depaiva@gmail.com Richard Garner, Macquarie University: richard.garner@mq.edu.au Ezra Getzler, Northwestern University: getzler (at) northwestern(dot)edu Kathryn Hess, Ecole Polytechnique Fédérale de Lausanne: kathryn.hess@epfl.ch Dirk Hoffman, Universidade de Aveiro: dirk@ua.pt Pieter Hofstra, Université d'Ottawa: phofstra (at) uottawa.ca Anders Kock, University of Aarhus: kock@math.au.dk Joachim Kock, Universitat Autònoma de Barcelona: kock (at) mat.uab.cat Stephen Lack, Macquarie University: steve.lack@mg.edu.au F. William Lawvere, State University of New York at Buffalo: wlawvere@buffalo.edu Tom Leinster, University of Edinburgh: Tom.Leinster@ed.ac.uk Matias Menni, Conicet and Universidad Nacional de La Plata, Argentina: matias.menni@gmail.com Ieke Moerdijk, Utrecht University: i.moerdijk@uu.nl Susan Niefield, Union College: niefiels@union.edu Robert Paré, Dalhousie University: pare@mathstat.dal.ca Kate Ponto, University of Kentucky: kate.ponto (at) uky.edu Jiri Rosicky, Masaryk University: rosicky@math.muni.cz Giuseppe Rosolini, Università di Genova: rosolini@disi.unige.it Alex Simpson, University of Ljubljana: Alex.Simpson@fmf.uni-lj.si James Stasheff, University of North Carolina: jds@math.upenn.edu Ross Street, Macquarie University: ross.street@mg.edu.au Tim van der Linden, Université catholique de Louvain: tim.vanderlinden@uclouvain.be R. J. Wood, Dalhousie University: rjwood@mathstat.dal.ca